## Collegio Carlo Alberto

# Sanctions and Incentives to Repudiate External Debt

Carlo de Bassa Scheresberg Edoardo Grillo Francesco Passarelli

> No. 505 July 2017

## Carlo Alberto Notebooks

www.carloalberto.org/research/working-papers

© 2017 by Carlo de Bassa Scheresberg, Edoardo Grillo and Francesco Passarelli. Any opinions expressed here are those of the authors and not those of the Collegio Carlo Alberto.

## Sanctions and Incentives to Repudiate External Debt<sup>\*</sup>

Carlo de Bassa Scheresberg<sup>†</sup> Edoardo Grillo<sup>‡</sup> Francesco Passarelli<sup>§</sup>

November 2, 2017

#### Abstract

Economic sanctions can be a double-edge sword. The International Community usually levies them in the attempt to foment discontent with a hostile regime. But sanctions may provoke reactions which are costly for the International Community.

This paper studies how sanctions shape a government's incentives to default on foreign debt. A default frees resources that the government can use to buy political support. It then becomes a defensive reaction aimed at restoring political stability.

Data covering the period 1950-2005 reveal that sanctions provide an almost 2 percentage point increase in default probability one year later. An external debt crisis that occurs after a country has been hit by a sanction is associated with a 13% point decrease in internal political turmoil.

This paper suggests that sanctions may trigger an external debt crisis, and, when this happens, their effectiveness in undermining the stability of the ruling regime is limited. A certain degree of caution is necessary in the use of sanctions, especially when the target is a highly indebted country.

<sup>\*</sup>We thank Yotam Margalit, Massimo Morelli, Ugo Panizza, Giacomo Ponzetto, Guido Tabellini, and participants in a seminar at CRENOS - Sassari, the 16th JLAGV conference, the 2017 EEA-ESEM Meeting - Lisbon, and the 2017 Bruneck Workshop in Political Economy for very useful comments.

<sup>&</sup>lt;sup>†</sup>George Washington School of Business. 2201 G St NW, Washington, DC 20052, United States. Email: carlo.debassa@gmail.com

<sup>&</sup>lt;sup>‡</sup>Collegio Carlo Alberto. Piazza Arbarello 8, Torino, 10122, Italy. Email: edoardo.grillo@carloalberto.org

<sup>&</sup>lt;sup>§</sup>Università di Torino and Università Bocconi. C.so Unione Sovietica 218 bis, 10134, Turin, Italy. Email: francesco.passarelli@unito.it

## 1 Introduction

Myanmar has been ruled by a military junta from 1962 to 2011. In 1988, the regime violently repressed demonstrations against economic mismanagement and in favor of a democratic transition. As a result, during the following decade US, EU, and several other countries levied economic sanctions against Myanmar. In 2002, the country defaulted on its external debt. Despite the efforts of the International Community, the regime remained in power until 2011.

Economic sanctions represent an important foreign policy tool, a middle course of action in-between diplomacy and war. As in the case of Myanmar, they are frequently imposed with the aim of impairing a hostile regime. However, despite their growing role in international disputes, their utility is yet a controversial issue.<sup>1</sup> Sanctions are costly, not only for the targeted country, but also for firms and workers in the sanctioning countries. The costs borne may not worth the benefits. Current literature does not provide clear-cut evidence that sanctions undermine the regime's political stability. Skeptics argue that they may even strengthen the popularity of the ruling government, promoting a nationalistic rhetoric.<sup>2</sup>

Moreover, as suggested by the Myanmar case, a specific link between sanctions and defaults may be at work. Defaulting on external debt may become more "politically acceptable" if it is advertised as a reaction to a sanction imposed by unfriendly foreign countries. It may also enable the government to divert resources from servicing the external debt to instrumentally buying internal support. In this case, the sanction would have limited impact on the regime's stability, while the cost borne by the International Community would be quite large.

Is this likely to happen? In a database, which collects information about sanctions, sovereign defaults, and political unrest in the period 1950-2005, we found that the chance to enter an external debt crisis was substantially higher in the subset of countries which were hit by a sanction some years earlier. We also found that sanctioned countries experienced higher internal political instability. However, political turmoil decreased in those countries if they defaulted on external debt some years later.

This paper studies the link between economic sanctions, debt crises and political turmoil. While existing literature argues that the threat of international sanctions acts as a deterrent for defaults, here we turn the issue around and study whether sanctions,

<sup>&</sup>lt;sup>1</sup>For a comprehensive view of the long debate over the utility of sanctions, cf. Haufber et al. (2009).

<sup>&</sup>lt;sup>2</sup>Following the sanctions levied by the US against Venezuela in 2011, the Venezuelan President Hugo Chavez portrayed this act as a "gringo aggression", arguing that it would reinforce "the nationalist and patriotic morale of Venezuela." Grossman, Manekin, and Margalit (2017) claim, and empircally test, that sanctions are likely to raise hostility against sanctioning countries.

imposed to a country for any reason (but unrelated to a default), would give a government more incentives to default on its sovereign debt.

We present a theoretical model in which the International Community levies a sanction that has the effect to exhaust the country economically, in the attempt to foment the population's discontent with the regime. The latter has to choose the income tax rate and how much to invest in activities to counteract the opposition. Tax proceeds are used also to service public debt, but the government can decide not to repay it, namely to default. Of course, a default is costly for the country, due to reputational losses and systemic consequences. A default, however, yields also some benefits. Once unburdened of the public debt, the government can lower taxes or invest more resources to fight opposition. Both these policies lower political instability and enhance the government's probability to survive in office. We show that when the country is hit by a sanction, such benefits may be larger, making the choice to default more attractive to the government. We further assume that, at the time the sanction is levied, financial markets form rational expectations about the probability with which a default occurs in equilibrium. In doing so, they anticipate the higher chance of a debt crisis following a sanction. Thus, they demand a risk premium rise on the government debt. This, in turn, further strengthens the government's incentive to enter a debt crisis.

The main implication of the model is that a default *following* a sanction dampens political instability. We find support in the data for this relationship. We regress political instability (as measured either by the frequency of riots or by a composite Conflict Index) against three dummies capturing (a) whether the country was under sanction, (b) whether it was in an external debt crisis, and (c) whether it was under a sanction and in an external debt crisis status, *provided that the former occurred one year before the latter*. We find that the first two dummies are not significantly associated to political instability, while the last interaction variable is significantly correlated with a decrease in the level of political unrest.

This evidence is consistent with the implications of the theoretical model. A sanction may trigger a *political* default on external debt which, at least partially, offsets the International Community's attempt to foment political instability and to pursue a regime change. Importantly, these political defaults can arise also when the *economic* fundamentals of the country would not lead to a debt crisis. Therefore, the default on external debt may backfire on foreign countries, especially those holding a substantial amount of debt. But, why are sanctions so frequently used? We propose two explanations. First, the International Community may have an interest in punishing the misconduct of a regime *per se*, independently of the sanction's effect on its chance of surviving in office. Second, the collective decision on whether to impose a sanction is realistically influenced by more powerful members. If these members own relatively small amounts of the country's debt, they will be relatively uninterested in the risk of triggering a default. This would lead to an excessive use of sanctions and to frequent debt crises occurring after sanctions.

The paper is organized as follows. Section 2 presents an overview of the related literature. Section 3 provides motivating evidence for our analysis. Section 4 introduces the theoretical model. Section 5 analyzes the equilibrium and presents theoretical predictions. Section 6 contains the empirical analysis. Section 7 concludes. The Appendix (Section 8) contains the proofs of propositions, and additional empirical evidence.

## 2 Related Literature

Given their widespread use as a tool to handle international disputes, economic sanctions have been independently studied by scholars in economics and political science. Most of the literature has focused on the reasons behind the decision to impose a sanction and on whether such a sanction is ultimately effective.<sup>3</sup> Dating back to Galtung (1967), economic sanctions are intended to favor the "political disintegration" of a leader's support within the targeted country. This "instrumental" view highlights how sanctions aim at the removal of a leader, whose interests are misaligned with those of the International Community (Marinov, 2005; Escribà-Folch and Wright, 2010).<sup>4</sup> In partial contrast with this idea, other scholars (e.g. Kaempfer and Lowenberg, 1988, 1992; Lacy and Niou, 2004) have pointed out that sanctions may also be imposed for "expressive" reasons, namely to take a moral or political stance against the leader's behavior and to send him a signal about one's resolve. In this paper, we allow both motives to be at play. Indeed, members of the International Community enjoy an *economic* benefit if the leader is overthrown but, independently, they also experience a *political* utility by using a sanction "expressively". Moreover, in line with our motivation, we enrich the utility function of the coercers by explicitly accounting for the negative consequences they may suffer if the targeted country defaults. The decision to impose a sanction may (efficiently) serve political-economic objectives or (inefficiently) result from a politically distorted collective decision.

There is controversy in the literature about the efficacy of sanctions as a tool to

<sup>&</sup>lt;sup>3</sup>See Baldwin (2000) for a general discussion of the political logic behind sanctions and Drezner (1999) and Hufbauer *et al.* (2009) for reviews of their scope and effectiveness.

<sup>&</sup>lt;sup>4</sup>Economic sanctions are then the "stick" in the hands of the international community to discipline a leader. The literature, however, has also investigated the "carrot-oriented" role of sanctions. For instance, Toke and Albornoz (2011) look at how changes in the flow of FDIs may help aligning the behavior of a leader with the interests of the international community.

discipline misaligned regimes. While Marinov (2005), Lektzian and Souva (2007) and Escribà-Folch and Wright (2010) conclude that sanction actually weaken authoritarian leaders, other scholars cast some doubts on this finding (e.g. Pape, 1997; Allen, 2008; Licht, 2017; Grossman et al., 2017). Some papers also point out that sanctions may decrease living standards within the targeted country (Weiss, 1997; Allen and Lektzian 2013; Neuenkirch and Neumeier, 2015) and increase the level or repression and violence (Wood, 2008; Peksen, 2009; Peksen and Drury, 2010; Hultman and Peksen, 2015). Our paper is close to those casting doubts on sanctions' effectiveness. Indeed, in our setting, the leader's decision to default on public debt backfires on foreign countries, offsetting, at least partially, their attempt to overthrow the leader.

To properly account for the instrumental view described above, we explicitly model the channel through which sanctions affect the leader's probability to survive in office. In this respect, we are close (among others) to Kaempfer *et al.* (2004), Wood (2008), De Mesquita and Smith (2009, 2010), Peksen and Drury (2010), and Oechslin (2014), which study how international sanctions may affect the behavior of an office-motivated leader. We differ from these papers as we focus on the use of a default on external debt as a reaction to the sanction.<sup>5</sup>

The relationship between international sanctions and sovereign defaults has been explored by the literature on "supersanctions" and "gunboat diplomacy" (e.g. Bulow and Rogoff, 1989a, 1989b; Weidenmier 2005; Mitchener and Weidenmier 2010).<sup>6</sup> This literature argues that international sanctions are imposed to induce debt repayment because of the severe and prolonged economic costs borne by the sanctioned country. In other words, defaults *cause* sanctions. Although in several instances this may have been the case, the link emphasized in the present paper goes in the opposite direction: sanctions *cause* defaults. We claim that a sanction may lead an office-motivated government to repudiate foreign debt in order to free resources that can be used to regain political support.

This paper is an instance of how political constraints may affect the choice to service public debt.<sup>7</sup> There is ample empirical evidence that political uncertainty is priced by financial markets (Citron and Nickelsburg, 1987; Brewer and Rivoli, 1997; Kohlscheen, 2007; van Rijckeghem and Weder, 2009). Political risk has also been incorporated in dynamic general equilibrium models (Cuadra and Sapriza, 2008 and Andreasen *et al.* 

<sup>&</sup>lt;sup>5</sup>We also differ from this literature because we model the revolt subgame as a global game (cf. Alkeson, 2000 and Edmond, 2013). See Morris and Shin (2003) for a review of static global games and Angelots *et al.* (2007) for the analysis of dynamic global games

<sup>&</sup>lt;sup>6</sup>See also Cole and Kehoe (1998) for the reputational costs that a default may cause.

<sup>&</sup>lt;sup>7</sup>For surveys on the extensive economic literature on sovereign defaults see Eaton and Fernandez (1995), Panizza, Sturzenegger and Zettelmeyer (2009) and Tomz and Wright (2013).

2016).<sup>8</sup> We contribute to this literature by explicitly modeling the link between sanction, revolt risk, and the probability to enter a default.

In our model, a default is effective in improving the government's probability to survive in office only if the cost borne by domestic citizens is not too high. This establishes a link with the literature investigating the costs of a default for the defaulting country (English 1996; Arellano 2008; Borensztein and Panizza 2009; Gelos *et al.* 2011; Yeyati and Panizza 2011; Cruces and Trebesch 2013; Gennaioli *et al.* 2014; Sandleris 2016). Finally, our paper is related to the finance literature on strategic defaults by households (e.g. Guiso *et al.* 2013) and corporations (e.g. Anderson and Sundaresan 1996; Mella-Barral and Perraudin 1997).

## 3 Some preliminary evidence

Our dataset collects information on 68 countries over the period 1950-2005, with a total of 3,091 country/year observations. It combines three different sources: the "Threats and Imposition of Sanction Database" on sanctions (henceforth, TISD); the Reinhart and Rogoff's Database on debt and financial crisis (henceforth RRD); the "Cross-National Time-Series Data Archive" on political institutions and political unrest (henceforth, CNTSDA). The Data Appendix provides a detailed description of these sources and precise definition of variables.<sup>9</sup> Table A1 in the Appendix reports the list of countries and the frequency of sanctions and external debt crisis for each of them.

In this dataset, sanctions occur quite frequently. Countries are recorded to be under sanction in more than 27% of the sample. Sovereign debt crises are less frequent (15.49% of the sample), and in most cases (92.7%) they involve external debt (see Tables A2-A3 in Appendix).

Table 1 shows that sanctioned countries are richer than countries under no sanction, while countries that default on external debt are poorer than non-defaulting countries. However, when it comes to political characteristics, sanctioned and defaulting countries are much more similar. The sanctioned countries are more likely to be ruled by a military junta and less likely to have an elective government, compared to countries under no sanction. They also experience a higher degree of internal political conflict (as measured by

<sup>&</sup>lt;sup>8</sup>On a complementary note, Amador, (2003) and Guembel and Sussman (2009) show that debt repayment may arise in the absence of default penalties for internal political reasons.

 $<sup>^{9}3,091</sup>$  is the number of observations for which data on all variables are available. In some of the regressions in Section 5, we supplement the dataset with public finance variables obtained from the World Bank World Development Indicators (henceforth, WDI). With the addition of this set of variables, the sample size shriks to 981 observations.

the frequency of riots and by the Conflict Index, which aggregates information on riots, strikes, political assassinations, guerrilla warfare and other measures of political unrest). Similar political patterns characterize countries under an external debt crisis: lower frequency of elective institutions and stronger political conflict, compared to countries that service public debt.<sup>10</sup>

### TABLE 1 HERE

Thus, when considered in isolation, being subject to a sanction and being in an external debt crisis are both associated with higher internal political instability. Interestingly, however, when the two conditions occur jointly, political instability decreases substantially. Table 2 tells us that being in an external debt crisis *and* under sanction is significantly associated with a lower incidence of riots and with a lower Conflict Index.<sup>11</sup>

#### TABLE 2 HERE

This evidence represents the main motivation of the theoretical model we present in the next section. Defaulting on external debt is the mechanism through which a country that has been hit by a sanction may regain internal political stability.

## 4 The Model

A country is ruled by an office-motivated head of state, Player H, (or the leader, terms which we use interchangeably) who enjoys a benefit v > 0 from keeping his position. The country has an outstanding amount of external public debt equal to B > 0. This debt is financed through a government bond, which must be repaid, together with interests, at the end of the game. Thus, the total cost of servicing the public debt is (1 + r) B, where r is the interest rate paid on the government bond.

<sup>&</sup>lt;sup>10</sup>All these differences are statistically significant at least at the 5% level. Not surprisingly, if we restrict the sample size to include additional variables, we further find that being in a default status is positively associated with a higher frequency of inflation crisis, lower reserves, higher stocks of external debt and higher payments on external debt (see Table A3 in Appendix). Similar patterns hold true for countries under sanction. Moreover, in this restricted sample, sanctions are still associated with greater political unrest, while the positive correlation between external debt crisis and political instability is weaker.

<sup>&</sup>lt;sup>11</sup>As shown in Table 2, the differences in our measures of politically instability are both statistically significant when we condition to the sanction status, while only the difference in riots is statistically significant when we condition on the external debt crisis status.

The population of the country is homogeneous in terms of income. We normalize income level to 1. Citizens belong to one of two possible groups, A and Z.

Individuals in group A are Activists. Each activist must independently choose whether to engage in a collective political attack against the leader in the attempt to overthrow him. The actual interpretation of such a choice may vary with the specific context we are analyzing. In autocratic regimes it often entails participating in a violent revolt against the dictator. In democratic regimes it may range from taking part in legal political movements or peaceful demonstrations to engaging in violent protests against the government. In either cases, the aim is weakening the political support of the ruling executive and thus favoring a parliamentary confidence crises.

Individuals in group Z, instead, do not directly take part in the revolt. To simplify the analysis, we assume that group A represents a small fraction of the total population, while group Z (and the country's population) has unit mass.

Citizens' gross income (hence, the country's GDP) is affected by sanctions imposed by the International Community. A sanction equal to  $\sigma \in [1, \bar{\sigma}]$  reduces the income of citizens to  $y(\sigma) = 1/\sigma$ . Thus,  $\sigma = 1$  implies no sanction while  $\bar{\sigma} > 1$  is the sanction's maximum level. One may think of  $1/\bar{\sigma}$  as the per capita GDP if the country is totally banned from the International Community.

Each sanctioning country within the International Community would draw an economic or political benefit from overthrowing the leader of the sanctioned country. But it would also bear a cost in the event of a default. Such cost is proportional to the amount of the country's external debt owned by the sanctioning country. Thus International Community members may have different views about how harsh the sanction should be. We assume that the level of sanction chosen by the International Community maximizes a weighted sum of members' utilities. Weights reflect their political influence within the International Community, capturing the idea that more powerful countries have a higher ability to affect the common decision in their favor (cf. Subsection 5.4 for additional details).

The utility of citizens in group Z depends on the sanction level,  $\sigma$ , and on the leader's decisions regarding taxation,  $\tau$ , and default,  $\delta$ :

$$u_Z(\sigma,\tau,\delta) = \frac{1-\tau}{\sigma} - \delta\ell,\tag{1}$$

In particular,  $\delta \in \{0, 1\}$  is a binary choice variable describing whether the country defaults on the outstanding debt ( $\delta = 1$ ) or not ( $\delta = 0$ ), and  $\tau \in [0, 1]$  is the proportional tax rate chosen by the leader to finance public expenditures (see subsection 5.2 below).  $\ell$  is the cost incurred by citizens if a default takes place.<sup>12</sup> As such,  $\ell$  may capture different factors, such as capital losses in citizens' portfolios, systemic effects on the financial sector, loss of international credibility, or any other utility loss associated with the socioeconomic turmoil a default may cause. We assume that  $\ell$  is uniformly distributed in the interval  $[0, \ell_H]$ . In particular,  $\ell$  is realized before the leader chooses whether to default or not (and it is observable to him), but after the sanction is determined.<sup>13</sup>

The regime is overthrown if and only if the fraction of activists who revolt exceeds a critical threshold. The latter positively depends on the utility of citizens in group Z,  $u_Z(\sigma, \tau, \delta)$ . Intuitively, when the discontent among citizens is high, the regime is less stable and, consequently, fewer activists are needed to overthrow it.

If an activist revolts and the revolt succeeds, he enjoys a payoff g > 0. Such payoff can be seen as the sum of a hedonic benefit, due to taking part in a successful political action, and a monetary payoff possibly related to the value of office, v. The idea is that those who participate in the political action overthrowing the regime not only are ideologically motivated, but they may also enjoy a material benefit from replacing the current ruling class. On the contrary, if the revolt fails each participating activist incurs a cost equal to the amount the leader invests in overthrow prevention,  $\pi \ge 0$ . One may think of  $\pi$  as the leader's investment in intelligence or riot police. In this interpretation, the activist's cost derives from a higher probability of being arrested or injured. The utility of an activist is thus given by:

$$u_A(\sigma,\tau,\delta,\pi) = \frac{1-\tau}{\sigma} - \delta\ell + g\Phi - \pi(1-\Phi), \qquad (2)$$

where  $\Phi$  is an indicator function that equals 1 if the activist participates in the revolt and it succeeds and 0 if he participates in it, but it fails. As we explain below, activists exhibit strategic complementarities in their behavior: the larger is the fraction of activists who revolt, the more each single activist has an incentive to revolt. This is because, as the revolt becomes larger in size, it also becomes more likely to succeed. Then, each activist faces a higher expected net benefit of revolting and this boosts his propensity to take part in the riot.

The investment in overthrow prevention,  $\pi$ , is financed through a proportional tax

<sup>&</sup>lt;sup>12</sup>We assume that the cost of defaulting,  $\ell$ , is independent of the level of sanctions. If we assumed, in line with Arellano, 2008, that this cost is decreasing in the level of sanctions ( $\ell(\sigma)$  with  $\ell'(\sigma) < 0$ ), there would be an even stronger incentive to default for the leader. All the insights from our anlaysis would then survive.

<sup>&</sup>lt;sup>13</sup>The randomness of  $\ell$  at the time the sanction is levied seems realistic. Moreover, it "smooths" the binary decision to default into a probabilistic event. The assumption of uniform distribution is made to simplify computations but, qualitatively, none of our results hinges on it. In what follows, we further assume that  $\ell_H$  is sufficiently high to guarantee that a strategic default occurs with probability strictly between 0 and 1 for any level of sanction  $\sigma \in [1, \bar{\sigma}]$ .

rate,  $\tau$ , levied on citizens' income. Tax revenues must also service public debt if no default occurs (i.e. if  $\delta = 0$ ):

$$\frac{\tau}{\sigma} = \pi + B \left( 1 + r \right) \left( 1 - \delta \right). \tag{3}$$

Given the sanction level,  $\sigma$ , and the interest rate, r, the leader chooses a policy vector  $(\tau, \pi, \delta) \in [0, 1] \times [0, 1] \times \{0, 1\}$  so as to maximize his probability to survive in office under a balanced budget constraint (3).

The interest rate r is an endogenous variable. It is determined by "financial markets" through a no-arbitrage condition against a risk-free interest rate,  $\tilde{r}$ . We say the *debt is priced* if there exists an interest rate r on the government bond that equates the expected return on the government bond (accounting for the risk of nonpayment) with the risk-free return:

$$(1+r)(1-F(\sigma | r)) = 1 + \tilde{r}.$$
(4)

 $F(\sigma | r)$  is the nonpayment probability, namely the probability with which Player H chooses to default when the sanction level is equal to  $\sigma$  and the interest rate is equal to r. Since the probability of a default depends on  $\sigma$ , the interest rate depends on the sanction level,  $r = R(\sigma)$ . If there is no r satisfying (4), we say the *debt is not priced* and we assume that a default happens with certainty. One might think of this as the case of a severe liquidity crisis. Thus, in our model, defaults can happen either because Player H decides so, or because risk premium is so high that default becomes inevitable. We refer to the former type of defaults as to *political defaults* and to the latter as to *economic defaults*.



Figure 1: Timing of the Model.

The timing of the events is the following (cf. Figure 1). In period 0, the International Community chooses the sanction level. In period 1, financial markets form rational expectations on the probability with which the country defaults given the level of sanctions  $\sigma$ ,

and interest rate r is determined. If the debt is priced, then in period 2 the leader chooses a policy vector  $(\tau, \pi, \delta)$ . If the debt is not priced, the default occurs with certainty and the leader chooses a vector  $(\tau, \pi)$ . In period 3, activists decide whether to revolt or not and, in period 4, the outcome of the revolt is determined.

## 5 The Analysis

This Section characterizes the equilibrium of the model. To ease the presentation, we proceed backward. First, we describe the revolt subgame. Then, we compute the optimal policy vector  $(\tau, \pi, \delta)$  and the interest rate,  $R(\sigma)$ . Finally, we analyze the choice of the sanction level by the International Community.

## 5.1 The Activists

At the beginning of period 3, each activist unilaterally decides whether to participate in a political attack against the leader (henceforth revolt).<sup>14</sup> The revolt succeeds and Player H is overthrown if and only if sufficiently many activists revolt. The revolt subgame among activists exhibits strategic complementarities: if more activists participate in the revolt, each activist has an individually higher incentive to revolt. A possible interpretation is that the hedonic feeling of contributing to a meaningful event is greater if the chance of success is higher. The complementarity is also on the cost side: the probability of being arrested or injured is smaller in a larger crowd.

To deal with equilibrium multiplicity that strategic complementarities may generate, we model the revolt as a global game (see Carlsson and van Damme, 1993, Morris and Shin, 1998, Atkeson, 2000, and Edmond, 2013). We assume that the fraction of activists needed to overthrow the regime is equal to  $u_Z(\sigma, \tau, \delta) + \eta$ . This is the citizens' degree of satisfaction with the regime. We refer to it as to the *regime stability*. Intuitively, a revolt succeeds more easily if it finds a larger support among citizens. This support is inversely related to their satisfaction with the current government, which depends both on their utility,  $u_Z(\sigma, \tau, \delta)$ , and on the realization of a random shock to the leader popularity,  $\eta$ . As such,  $\eta$  captures those factors which do not enter in the government budget constraint (3), but they still affect citizens' satisfaction with the regime (e.g., their affection to the leader or the effects of his rhetoric on them). The randomness of  $\eta$  is realistic, since the precise impact of these factors depends somehow on chance. Each activist shares a common prior that  $\eta$  is uniformly distributed in the interval  $[\eta, \overline{\eta}]$ . Although  $\eta$  is realized

<sup>&</sup>lt;sup>14</sup>The coordination mechanism in a revolt (either peaceful or violent) is sufficiently general to be illustrative of other forms of collective political action against the regime.

at the beginning of period 3, activists may not be perfectly informed about its realization. They may have some private information, but it may be biased and it may fail to represent the actual support enjoyed by the leader. For instance, an activist may be able to assess the leader's popularity in his neighborhood or within his social network, but not in others. Formally, each activist *i* observes an independent signal  $\varepsilon_i$  uniformly distributed in the interval  $\left[\eta - \frac{1}{2\psi}, \eta + \frac{1}{2\psi}\right]$ .<sup>15</sup> After receiving their signals activists form their posteriors: a distribution over the leader's popularity that is conditional on the signal. Assumption 1 below guarantees that activists who receive sufficiently low (resp., high) signals, want to revolt (resp., not to revolt) independently of what other activists do. This assumption is standard in the global games literature and yields equilibrium uniqueness. To state it formally, let  $\overline{u} = 1$  and  $\underline{u} = -\ell_H$  be respectively the highest and lowest utility citizens can experience in the game.

Assumption 1 Revolting is dominant (resp., dominated) if the popularity shock is sufficiently low (resp., high):  $\overline{u} + \underline{\eta} < 0$  and  $\underline{u} + \overline{\eta} > 1$ . Equivalently,  $\overline{\eta} > 1 + \ell_H$  and  $\underline{\eta} < -1$ .

Proposition 1 states that the revolt subgame has an essentially unique equilibrium and characterizes the survival probability of the leader. The proof follows the same argument as in Morris and Shin, 2002 (see Appendix for details).

**Proposition 1** Under Assumption 1, there are unique  $\varepsilon^*$  and  $\eta^*$  such that in any equilibrium of the revolt subgame, (i) an activist revolts if and only if  $\varepsilon_i \leq \varepsilon^*$ , and (ii) the leader is overthrown if and only if  $\eta \leq \eta^*$ .<sup>16</sup> Furthermore, the leader's survival probability is given by

$$S(\sigma,\tau,\pi,\delta) = \Pr\left\{\eta \ge \eta^*\right\} = 1 - \frac{1}{\overline{\eta} - \underline{\eta}} \left(\frac{g}{g + \pi} - u_Z(\sigma,\tau,\delta)\right).$$
(5)

Comparative statics are straightforward. The leader's chance to survive is larger when activists draw lower utility from overthrowing him (lower g) and when citizens' utility is higher (larger  $u_Z$ ). On next sections we discuss the effect of the policy variables,  $\sigma, \pi, \tau$ , and  $\delta$ , on this survival probability.

<sup>&</sup>lt;sup>15</sup>By construction,  $\varepsilon_i$  can take values in the interval  $\left[\underline{\eta} - \frac{1}{2\psi}, \overline{\eta} + \frac{1}{2\psi}\right]$ . The posterior belief after signal  $\varepsilon_i$  can thus be determined through Bayes rule as follows. If  $\varepsilon_i \in \left[\underline{\eta} + \frac{1}{2\psi}, \overline{\eta} - \frac{1}{2\psi}\right]$ , then  $\eta \mid \varepsilon_i \sim U\left[\varepsilon_i - \frac{1}{2\psi}, \varepsilon_i + \frac{1}{2\psi}\right]$ . Instead, if  $\varepsilon_i > \overline{\eta} - \frac{1}{2\psi}$  (resp.,  $\varepsilon_i < \underline{\eta} + \frac{1}{2\psi}$ ), then  $\eta \mid \varepsilon_i \sim U\left[\varepsilon_i - \frac{1}{2\psi}, \overline{\eta}\right]$  (resp.,  $\eta \mid \varepsilon_i \sim U\left[\eta, \varepsilon_i + \frac{1}{2\psi}\right]$ ).

 $<sup>\</sup>eta \mid \varepsilon_i \sim U \left[\underline{\eta}, \varepsilon_i + \underline{1}_{2\psi}\right]$ ). <sup>16</sup>If  $\varepsilon_i = \varepsilon^*$ , activist *i* is indifferent between revolting or not. We break this indifference assuming that he revolts for sure.

## 5.2 The Leader

In period 2, Player *H* takes the level of sanctions  $\sigma$  as given and chooses the policy vector  $(\tau, \pi, \delta) \in [0, 1] \times [0, 1] \times \{0, 1\} \equiv C$  to maximize his probability to survive in office,  $S(\sigma, \tau, \pi, \delta)$ , subject to the government budget constraint. Thus, if the debt is priced, the budget constraint is (3), and the leader solves the following problem:

$$\max_{(\tau,\pi,\delta)\in C} \upsilon S\left(\sigma,\tau,\pi,\delta\right) \text{ s.t. } \quad \frac{\tau}{\sigma} = \pi + B\left(1+r\right)\left(1-\delta\right). \tag{6}$$

If the debt is not priced, the leader cannot service the debt.  $\delta$  is no longer a choice variable, and the budget constraint is  $\frac{\tau}{\sigma} = \pi$ . Thus, the leader's problem becomes:

$$\max_{(\tau,\pi)\in[0,1]^2} \upsilon S\left(\sigma,\tau,\pi,1\right) \text{ s.t. } \quad \frac{\tau}{\sigma} = \pi.$$
(7)

First, suppose the debt is priced. The investment in overthrow prevention,  $\pi$ , has two effects on survival probability (cf. expression (5)). On the one hand, survival probability increases because a higher  $\pi$  (e.g. stronger intelligence or police) makes it less worthwhile to run the risk of revolting. Less activists are sucked into the revolt, thus overthrow becomes less likely. On the other hand, a higher investment in overthrow prevention implies higher taxes and lower  $u_Z(\sigma, \tau, \delta)$ . Citizens are less satisfied with the government and more prone to support the revolt by the activists.

Similarly, the decision to default has a two-fold effect on the leader's probability to survive in office. On the one hand, citizens suffer a cost equal to  $\ell$  and their discontent with the current government goes up, dampening survival probability — see (1). On the other hand, a default eliminates the cost of servicing the public debt, thus enabling a tax reduction and an increase in  $u_Z(\sigma, \tau, \delta)$ , which boosts the leader's survival probability.

Importantly, the optimal policy vector further depends on whether constraint  $\tau \leq 1$  binds or not — recall that  $\tau \in [0, 1]$ . If it binds, the choice of  $\pi$  and  $\delta$  is shaped by an additional trade-off. If the leader chooses to service the debt ( $\delta^* = 0$ ), then the investment in overthrow prevention must be chosen residually after setting taxation at the highest possible level ( $\tau^* = 1$ ). Thus,  $\pi^*$  is lower than the level that would be chosen in the absence of constraint  $\tau \leq 1$ . On the contrary, if the leader insists on choosing overthrow prevention based on the trade-off described above, he must renege on the public debt ( $\delta^* = 1$ ). In what follows, we assume that these two cases are exhaustive for the equilibrium analysis. This amounts to assuming that the tax base is high enough to either set overthrow prevention at its first-best level in the absence of interest payments

or to service public debt in the absence of default risk.<sup>17</sup>

The optimal policy vector,  $(\tau^*, \pi^*, \delta^*)$  solves the trade-offs highlighted above. It is a function of the sanction level  $(\sigma)$ , the realization of the default cost  $(\ell)$ , and the interest rate (r). However, as we clarify below, the interest rate is itself a function of  $\sigma$ . Thus we can write  $\tau^* = T(\sigma, \ell \mid r), \pi^* = P(\sigma, \ell \mid r)$ , and  $\delta^* = D(\sigma, \ell \mid r)$ .

First-best level of overthrow prevention solves (6) ignoring constraint  $\tau \leq 1$ . In particular, such level is equal to  $\sqrt{g} - g \geq 0$  (cf. Lemma 4 in Appendix). The inequality tells us that the leader invests a (weakly) positive amount in revolt prevention only if the activists' benefit of a successful revolt is not too high (g < 1). If g > 1, the non-negativity constraint on  $\pi$  yields a corner solution: the leader finds it optimal not to invest at all.<sup>18</sup> We rule out this corner solution with the following assumption:

#### **Assumption 2** The activists' benefit of a successful revolt is not too high: g < 1.

Based on Assumptions 1 and 2, Lemma 4 in the Appendix characterizes the equilibrium policy vector chosen by the leader. Moreover, exploiting the randomness in the cost of defaulting  $\ell$ , we can compute the probability of a political default, as it is assessed by the International Community and by financial markets. Let this probability be  $F(\sigma | r) = \int_{0}^{\ell_{H}} D(\sigma, \ell | r)$ . The following proposition states an important result.<sup>19</sup>

**Proposition 2** For any value of the interest rate  $r \geq \tilde{r}$ , the probability of a political default,  $F(\sigma | r)$ , is increasing in  $\sigma$ .

To understand why a harsher sanction is more likely to trigger a default, first observe that the leader's net benefit from defaulting is the sum of three separate forces.<sup>20</sup> First, a default enables to save (1 + r) B and reduce taxation, which makes citizens more satisfied. Revolts become more difficult, thus survival probability improves. Second, if the constraint  $\tau \leq 1$  binds, a default frees resources that can now be invested to reach the optimal level of overthrow prevention. The leader can spend more money in, say, riot police, with a positive effect on his probability to survive in office. Third, a default yields a disutility for the citizens equal to  $\ell$ . Citizens are less satisfied, with a negative impact on the leader's survival probability. An increase in the sanction level,  $\sigma$ , does not affect

<sup>&</sup>lt;sup>17</sup>Formally,  $1/\bar{\sigma} \ge \max\left\{\sqrt{g} - g, B\left(1 + \tilde{r}\right)\right\}$ . This implies that, absent any strategic reaction by an office-motivated leader, sanctions cannot automatically lead to a default.

<sup>&</sup>lt;sup>18</sup>Intuitively, if g > 1, the activits' benefit is so high that in order to discourage revolts the leader should "buy" the citizens' support with a negative  $\pi$ , which in turn implies negative taxes (hence, positive transfers). In the interest of realism, we restrict the investment in overthrow prevention,  $\pi$ , to be positive and we ignore this case.

<sup>&</sup>lt;sup>19</sup>The proof in the Appendix contains details about the analytical derivation.

 $<sup>^{20}</sup>$ See (15) in the Appendix for a formal characterization of these forces.

the first and third force, while it amplifies the second positive force. Indeed, harsher sanctions shrinks citizens' GDP per capita,  $\frac{1}{\sigma}$ , and, consequently, the country tax base. This reduction in the tax base makes the tax rate constraint,  $\tau \leq 1$ , more likely to be binding. Then, a default becomes more appealing to the leader.

Turning our attention to the case in which the debt is not priced, notice that in these situations, servicing the debt is financially not sustainable. Thus, an economic default occurs and the budget constraint becomes  $\frac{\tau}{\sigma} = \pi$ . Then, the leader chooses an amount of overthrow prevention equal to  $\pi^*$  and a level of taxation equal to  $\sigma\pi^*$ .

## 5.3 The Markets

In period 1, financial markets form rational expectations on the probability of a default,  $F(\sigma | r)$ , and they set the interest rate r on the government bond accordingly. The absence of arbitrage opportunities implies that the expected gross return on the government bond must equal the safe gross return that an investor can receive by investing in the risk-free asset. Formally:

$$(1+r)(1-F(\sigma | r)) = 1 + \tilde{r}.$$
(8)

The Appendix shows that an increase in r increases both the yield on the government bond, 1+r, and the probability of a political default,  $F(\sigma | r)$ . It also shows that equation (8) may admit up to two solutions. When such multiplicity arises, financial markets are indifferent between two values of the interest rate. For the sake of simplicity, we assume that in this case they pick the lowest value of r, which we denote with  $R(\sigma)$ .<sup>21</sup> Obviously,  $R(\sigma) \geq \tilde{r}$ .

For some values of  $\sigma$ , equation (8) admits no solutions. In this case the debt is not priced. This happens when the amount of outstanding debt is so high that, independently of the level of the interest rate, setting taxation in order to service this debt would generate too much discontent among the population. Then, the leader would always choose not to service the debt.

The following Proposition states that debt is likely to be priced when defaulting generates large expected turmoil costs (high  $\ell_H$ ), or when the risk-free bond yields a low return (small  $\tilde{r}$ ). Here is the intuition. As explained in the previous section, a high turmoil cost decreases the leader's net benefit of defaulting (see the discussion after Proposition 2).

 $<sup>^{21}</sup>$ The choice of the lowest interest rate solving (8) can be rationalized assuming that, due to some unmodelled factors, international investors, keeping expected returns constant, prefer to pick the interest rate associated with the lowest risk of default. For instance, this could happen if some investors fear the global financial turmoil that a default could generate and the consequence that this could have on them.

Then, markets are more likely to price the debt. Likewise, when the alternative risk-free investment is unattractive (low  $\tilde{r}$ ), markets price the debt even if the default probability is relatively high. Furthermore, the proposition also states that, when debt is priced and r is endogenously determined, the default probability is increasing in the sanction level.

**Proposition 3** The debt is priced if and only if its amount B does not exceed a threshold  $\overline{B} := \ell_H / [4(1+\tilde{r})]$ . If debt is priced, the interest rate,  $R(\sigma)$ , is a continuous function of the sanction. Moreover, there exists a  $\hat{\sigma}$  such that  $R(\sigma)$  is constant for  $\sigma \leq \hat{\sigma}$  and increasing for  $\sigma > \hat{\sigma}$ . As a result, the probability of a political default,  $F(\sigma | R(\sigma))$ , is increasing in  $\sigma$ .

Since the return on the government bond in determined by (8), we can totally differentiate such equation and conclude that  $d((1 + R(\sigma))(1 - F(\sigma | R(\sigma))))/d\sigma = 0$ . In other words, if the bond is priced and we marginally change the sanction level, the change in the yield on the government bond,  $R(\cdot)$ , and in the default probability,  $F(\cdot)$ , fully offset each other in order to preserve no-arbitrage.

Having endogenized the yield on the government bond, we are ready to define the leader's expected survival probability,  $\hat{S}$ , as it is assessed by the International Community in period 0. By the assumption on the distribution of  $\ell$ , we get:

$$\hat{S}(\sigma) = \int_{0}^{\ell_{H}} \frac{S(\sigma, T(\sigma, \ell \mid R(\sigma)), P(\sigma, \ell \mid R(\sigma)), D(\sigma, \ell \mid R(\sigma)))}{\ell_{H}} d\ell$$
(9)

Expected survival probability is a function of  $\sigma$ . This is the case because, by Proposition 1, the probability to survive in office is a function of the policy vector and of the yield on the government bond, which in turn all depend on  $\sigma$ . Proposition 4 below states that  $\hat{S}(\sigma)$  is decreasing in  $\sigma$ . This means that in this model, the traditional argument underlying the use of sanctions applies: sanctions weaken the leader and favor his overthrowing.

**Proposition 4** The expected survival probability of the leader,  $\hat{S}(\sigma)$ , is decreasing in the sanction level.

To understand Proposition 4, first suppose that the equilibrium policy vector,  $(\tau, \pi, \delta)$ , and the yield on the government bond, r, do not depend on the sanction level. In this case, by equations (1) and (5), an increase in the sanction level has only a "direct" effect: it lowers GDP per capita, which in turn reduces citizens' satisfaction and decreases the leader's survival probability. Furthermore, observe that (5) also implies that the survival probability of the leader increases with the investment in overthrow prevention,  $\pi$ , and decreases with the tax rate,  $\tau$ , and with the decision to default,  $\delta$ . Finally, since the service of debt is financed through taxation, an increase in the interest rate also leads to an increase in the tax rate and to a decrease in the leader's probability to survive in office.

Now, let us account for the fact that the policy vector and the yield on public bonds are themselves affected by the level of sanctions. Then, besides the direct impact on the GDP per capita described above, an increase in the sanction level has additional "indirect" effects. First, it leads to an increase in the yield  $R(\sigma)$  (cf. Proposition 3) and to a related increase in the tax rate, which lowers the leader's survival probability. Second, a harsher sanctions lowers the tax base of the country, shrinking the resources the leader can spend in overthrow prevention. This also lowers the leader's survival probability. The leader may offset this latter effect by defaulting more often. Importantly, Proposition 4 states (and Appendix proves) that, independently of how frequently the leader reacts with a political default, the overall effect of an increase in the sanction level is always a decrease his expected survival probability.

## 5.4 The International Community

At the beginning of period 0, the International Community chooses the sanction level taking all economic and political consequences into account. Specifically, it computes the interest rate that markets will set in equilibrium and it forms rational expectations about the event of a sovereign default and the event of an overthrowing in the targeted country.

Let the International Community be a set  $\Gamma = \{1, ..., n\}$  of member countries. Realistically, these countries have heterogeneous interests, not only in overthrowing the regime, but also in the repayment of the debt. On top of that, they may differ in their political influence when it comes to making a common decision about sanctions.

Suppose each country  $j \in \Gamma$  enjoys a political and economic benefit  $k_j$  from overthrowing the leader from office but it also bears a cost if the country defaults. Benefits derive from the improvement in economic and diplomatic relationships that occurs if a new friendly government replaces the current regime. Moreover, independently of the effect on regime survival, sanctions often yield political benefits per se. Indeed, countries may find themselves in the position of punishing a hostile regime, as a politically meaningful action showing that the misconduct of the targeted regime cannot be tolerated. Such "expressive" return from punishment, call it  $g_j$ , positively depends on the sanction but is independent of whether the regime survives or not:  $g_j = G_j(\sigma)$ , with  $G'_j > 0$ . For simplicity, hereafter we assume that benefits are the same for all countries:  $k_j = k$  and  $G_j(\sigma) = G(\sigma)$  (for all j).<sup>22</sup>

 $<sup>^{22}</sup>$ As discussed in Section 2, this expressive role of sanctions has been stressed by the literature. Nonethelss, our qualitative results would go through even if we shut down this channel, by setting

Let  $b_j$  be the amount of the target country's debt held by country j. The cost borne by j in case of a default includes both the loss  $b_j(1 + r)$  that occurs when the debt is not serviced, and a cost  $c_j$  capturing all other negative externalities the default may have on country j's economy (e.g., domino effects on the domestic financial system, confidence disruption, higher instability, reduction in trading flows with the defaulting country,...). In this interpretation, we assume that  $c_j$  is positively related to the amount of debt:  $c_j = C(b_j)$ , with C' > 0 and  $\bar{c} = \frac{1}{n} \sum_{j=1}^n c_j$ .

Summing up, if a sanction  $\sigma$  is levied, country j's expected utility is given by

$$v\left(\sigma \mid b_{j}\right) = G(\sigma) + k \cdot \left(1 - \hat{S}\left(\sigma\right)\right) - C(b_{j}) \cdot F\left(\sigma \mid R\left(\sigma\right)\right) + b_{j} \cdot \left(1 + \tilde{r}\right)$$
(10)

The first term in the LHS is the "expressive" return from punishing the regime. The second one is the expected "instrumental" return from overthrowing the leader. The third one is the expected cost associated with a default. The fourth term is the expected value of the bonds,  $b_j \cdot (1+r)(1-F(\sigma | r)) = b_j \cdot (1+\tilde{r})$ . This equality derives from the no-arbitrage condition (8). It shows that such term is independent of the sanction because the bond yield automatically adjusts to include the default risk premium.

The normative benchmark is a sanction,  $\sigma^{o}$ , that maximizes the following Benthamite social welfare function,

$$W(\sigma \mid b) = \sum_{j \in \Gamma} v(\sigma \mid b_j), \qquad (11)$$

where  $b = \{b_1, ..., b_n\}$ . The level  $\sigma^o$  would be the sanction chosen by a social planner which is benevolent towards all the International Community's members. Eventually, this planner is unaffected by the political power of any member. At  $\sigma^o$  the average marginal benefit from punishing the country and increasing the chance to overthrow the leader,  $dG/d\sigma - k \cdot d\hat{S}(\sigma)/d\sigma$ , equals the average marginal cost of a higher exposure to the risk of default,  $\bar{c} \cdot dF(\sigma)/d\sigma$  (see Appendix for details).

Realistically, however, powerful countries may have the ability to influence the common decision in their favor. Let  $\phi_j$  parametrize country j's "de facto" political power within the International Community. This parameter captures the diplomatic, military and institutional factors that make a country more or less influential in the decisions regarding sanctions. Without loss of generality, we normalize total political power to one, namely  $\sum_{j=1}^{n} \phi_j = 1$ . To account for power heterogeneity, we assume that the International Community chooses a "politically distorted" sanction,  $\sigma^*$ , which maximizes a "distorted" Benthamite social welfare function where each country's utility function is

 $G\left( \sigma \right) \equiv 0.$ 

weighted by that country's political power. Let this distorted social welfare function be,<sup>23</sup>

$$\tilde{W}(\sigma \mid b, \phi) = \sum_{j \in \Gamma} \phi_j v(\sigma \mid b_j), \qquad (12)$$

with  $\phi = {\phi_1, ..., \phi_n}$ . The next proposition advances the idea that, compared to the normative benchmark, the sanction is too harsh (resp., too mild) if on average the powerful countries are relatively unexposed (resp., exposed) to the default risk.

**Proposition 5** Let the unweighted average cost of a default be  $\bar{c} = \frac{1}{n} \sum_{j=1}^{n} C(b_j)$ , and let  $\bar{c}_{\phi} = \sum_{j=1}^{n} \phi_j C(b_j)$  be the "power-weighted average cost". i) If  $\bar{c}_{\phi} \leq \bar{c}$  (resp.,  $\bar{c}_{\phi} \geq \bar{c}$ ) then the equilibrium sanction  $\sigma^*$  is socially too harsh,  $\sigma^* \geq \sigma^o$  (resp., too mild,  $\sigma^* \leq \sigma^o$ ); Assume the relation between cost and debt is linear,  $C(b_j) = \alpha + \beta b_j$ , ( $\beta > 0$ ). Let  $\bar{b}_{\phi} = \sum_{j=1}^{n} \phi_j b_j$  be the "power-weighted average amount of debt", and let  $\bar{b} = \frac{B}{n}$  be the

ii) If  $\bar{b}_{\phi} \leq \bar{b}$  (resp.,  $\bar{b}_{\phi} \geq \bar{b}$ ) then the sanction  $\sigma^*$  is socially too harsh (resp., too mild).

unweighted average.

This proposition points out that the source of political distortion in the choice of the sanction level is the uneven distribution of power within the International Community. When the more powerful members incur lower (resp., higher) costs in sanctioning a regime, the power-weighted average cost is smaller (resp., larger) than the unweighted average. Then the sanction is harsher (resp., milder) than the optimal level. Statement ii) points at the role of debt shares. The countries with lower amount of debt in their portfolios prefer harsher sanctions because they bear lower costs in case of default. If these countries are also the most powerful ones, the sanction will be too harsh (and vice versa). Specifically, if the default cost borne by each lender is linearly correlated to its amount of debt, then the condition for a "politically distorted" sanction becomes particularly simple: the sanction is too harsh (resp., too mild) if the power-weighted average debt share is smaller (resp., larger) than the unweighted average.

Summing up, a key message emerges from our theoretical analysis. Although sanctions shorten a regime's survival expectancy, they also increase the probability of a political default because the leader may use this political tool to restore some of his popularity. In our model, this is especially the case when imposing a sanction is an admonitory act with high political significance, or when the most powerful countries have little to lose from a default. In such situations the International Community, or at least its most powerful

 $<sup>^{23}</sup>$ Passarelli and Tabellini (2017) have a similar functional form to account for the influence of politically organized groups engaging in protests against the government.

members, may find it profitable to trade an increase in the risk of default against the political return of an admonitory punishment.

## 6 Empirical Evidence

Defaulting on external debt may represent the regime's optimal reaction to a sanction. An empirically testable prediction of our theoretical model concerns timing: external debt crisis are likely to occur *after* sanctions. In order to test this prediction, we regress the external debt status of a country over the existence of a sanction against that country in the previous year. In order to account for the possibility that some sanctions were imposed against the country *because of* a previous debt crisis, we consider only those observations for which the country was not in an external default status in the two previous years. We further include a set of political and economic controls as well as country and year fixed effects. Results are reported in Table 3.

### TABLE 3 HERE

Sanctions are positively and significantly associated with the entry in an external debt crisis one year later, providing more than a 1.5% percentage point increase in default probability. This evidence is consistent with Proposition 3 in the theoretical model. It holds true independently of whether economic and political variables are included among the controls (Columns 1, 2 and 3) or whether we consider the restricted sample for which information on public finance variables are included (Column 4).<sup>24</sup>

Our data suggest that sanctions are likely to occur before defaults. This evidence is consistent with the causal link that our theoretical model puts forward (namely, sanctions cause external debt crises), but it is also consistent with an alternative explanation claiming that sanctions may represent preemptive punishments against governments that are expected to default on foreign holders. In this case, the casual link would go in the opposite direction. In order to account for this alternative explanation and provide further evidence in support of our model, in the Appendix we show that the positive correlation between lagged sanctions and debt crisis is robust to different specifications of the temporal lag (see Tables A5-A7). Thus, if the alternative explanation were true, it would imply

<sup>&</sup>lt;sup>24</sup>Not surprisingly, richer countries (as measured by GDP per capita) and less indebted countries (as measured by the stock of external debt in percentage of GNI) are less likely to enter an external debt crisis.

that the International Community is able to predict that a country will default several years ahead, and punish that country quite long beforehand, which we deem unlikely.

In the Appendix (see Table A8), we ask whether there is a significant correlation between external debt crises starting at time t and sanctions occurring at t + 1 or t + 2, but we do not find evidence. Thus, there is no support in our data for the "gunboat diplomacy" explanation, which argues that sanctions typically represent the International Community's reaction to a default.

A second testable prediction of the model is that a default following a sanction dampens political instability and positively affects the regime's popularity. Thus one should observe lower political instability in countries that default after a sanction than in countries that do not default after being hit by a sanction. In order to test this implication, we regress measures of political instability against three dummies capturing (i) whether the country was under sanction in the previous two years, (ii) whether the country was in an external debt crisis in the previous two years, and (iii) whether in the two previous years the country was jointly in a sanction and in an external debt crisis, provided that the latter occurred one year *after* the former. This last interaction variable captures how the default decision correlates with the regime's stability when the country has been targeted by a sanction. A negative value of the coefficient would tell us that defaults following sanctions have a dampening effect on internal political tensions, as predicted by the model.

We consider two different measures of political instability: a dummy for the occurrence of riots and the Conflict Index defined in Section 3. For each of these measures we run two separate regressions: the first one uses the full sample and includes a set of political and economic controls as well as country and year fixed effects; the second one includes all previous variables plus additional public finance variables and it is thus restricted to the smaller sample for which all these information are available. Results are reported in Table 4.

#### TABLE 4 HERE

Columns (1)-(4) show that the interaction term between sanctions and external debt crises is associated with a decrease in the incidence of riots and in the Conflict Index. In particular, the average incidence of riots decreases by more than 13 percentage points, a quite large effect (see columns (1) and (3)). These two patterns are statistically significant respectively at 5% and 10% level. The estimated coefficients are even larger if we consider the restricted sample for which information on public finance variables are available (cf. columns (2) and (4)).<sup>25</sup> As expected, the data also reveal that political instability tends to increase in adverse economic conditions (lower GDP per capita growth).

## 7 Conclusion

In this paper we studied the relationship between sanctions and defaults. The common view in the current literature is that sanctions are imposed to punish insolvent countries or to induce debt repayment. According to this literature, defaults cause sanctions. Here, we advanced an alternative explanation: a default is a defensive strategy aimed at regaining political support after a sanction. The main mechanism is simple: sanctions make it harder for the government to gather support and defaults relax the government's budget constraint. Thus, our explanation reverts the causal relationship between sanctions and defaults.

We did not find a clean instrument to establish causality. However, our explanation appears to be more in line with empirical evidence than alternative explanations. We found that typically sanctions happen before defaults: the association between sanctions and lagged defaults is statistically significant, while there is no significant correlation between defaults and lagged sanctions.

Our empirical evidence on the link between sanctions, defaults, and political instability leads to the following considerations. Sanctions per se do not have a clear effect on political instability. This result is consistent with our theoretical conjecture that the International Community's decision to impose a sanction is frequently subject to political distortions. Thus the sanction can be too harsh or too mild. As a result, the effect on the leader's stability is unclear.<sup>26</sup> However, the picture is different when it comes to defaulting on foreign debt. When the country in under a sanction, a default is significantly associated with a substantial reduction in the degree of political conflict. This result suggests, and this is our second consideration, that a sanction may give the government the pretext to default on foreign debt, with the final goal of regaining political support.

While the general evaluation of the political effects of sanction is yet a controversial issue, this paper suggests that sanctions may trigger an external debt crisis, and, when this happens, their effectiveness in undermining the stability of the ruling regime may be limited.

 $<sup>^{25}</sup>$ Table A9 in the Appendix shows that this last pattern holds true if we increase the temporal lag, although the statistical significance is somehow reduced.

<sup>&</sup>lt;sup>26</sup>The evidence is also consistent with a lack of consensus in the empirical literature about the effects of sanctions on the governments' popularity and stability.

## 8 Appendix

## 8.1 **Proof of Propositions**

### **Proof of Proposition 1.**

Fix a level of sanctions  $\sigma$  and a policy vector  $(\tau, \pi, \delta)$ .. Pick any equilibrium of the revolt subgame and let  $\alpha(\varepsilon_i)$  be the fraction of activists who attack after they receive signal  $\varepsilon_i$ . Define  $Q(\eta, \alpha) = \psi \int_{\eta - \frac{1}{2\psi}}^{\eta + \frac{1}{2\psi}} \alpha(\varepsilon_i) d\varepsilon_i$ ; thus,  $Q(\eta, \alpha)$  is the overall fraction of activists who revolt if the realization of the popularity shock is  $\eta$ . Finally, let  $W(\alpha) = \{\eta : Q(\eta, \alpha) \ge v + \eta\}$  be the set of realizations of the popularity shock for which the revolt is successful.

Thus, we can summarize the payoff of each activist in the following table

	$\eta \in W\left(\alpha\right)$	$\eta \notin W\left(\alpha\right)$
Revolt	g	π
Not Revolt	0	0

As a result, the expected payoff of an activist who observes signal  $\varepsilon_i$  and decides to revolt is given by:

$$w\left(\varepsilon_{i},\alpha\right) = \int_{\eta \in W(\alpha)} g dG\left(\eta \mid \varepsilon_{i}\right) - \int_{\eta \notin W(\alpha)} \pi dG\left(\eta \mid \varepsilon_{i}\right),$$

where  $G(\eta | \varepsilon_i)$  denotes the conditional cdf of  $\eta$  given signal  $\varepsilon_i$ . On the contrary, if the activist does not participate in the revolt, he gets a safe payoff equal to 0. Since all activists are identical, in equilibrium:

$$\alpha(\varepsilon_i) = \begin{cases} 0 & \text{if } w(\varepsilon_i, \alpha) < 0; \\ x \in [0, 1] & \text{if } w(\varepsilon_i, \alpha) = 0; \\ 1 & \text{if } w(\varepsilon_i, \alpha) > 1. \end{cases}$$

Let  $\iota_k$  be a cutoff strategy that prescribe to revolt if  $\varepsilon_i$  is below k and not to revolt if  $\varepsilon_i$  is above k. The next two lemmata proves some useful properties of  $w(\varepsilon_i, \alpha)$ .

**Lemma 1** If  $\alpha_1(\varepsilon_i) \geq \alpha_2(\varepsilon_i)$  for every  $\varepsilon_i$ , then  $w(\varepsilon_i, \alpha_1) \geq w(\varepsilon_i, \alpha_2)$  for every  $\varepsilon_i$ . **Proof.** If  $\alpha_1(\varepsilon_i) \geq \alpha_2(\varepsilon_i)$  for every  $\varepsilon_i$  (namely, for every signal  $\varepsilon_i$ , a higher fraction of activist revolts under  $\alpha_1$  than under  $\alpha_2$ ), we know that for every  $\eta$ ,  $Q(\eta, \alpha_1) \geq Q(\eta, \alpha_2)$ . As a result,  $W(\alpha_1) \supseteq W(\alpha_2)$ . Thus, Player H is more likely to be overthrown if the revolting fraction of activists is given by  $\alpha_1(\cdot)$  instead of  $\alpha_2(\cdot)$ . The statement of the lemma follows from the definition of  $w(\varepsilon_i, \alpha)$  and the fact that  $g > 0 > -\pi$ . **Lemma 2**  $w(k, \iota_k)$  is continuous and strictly decreasing in k. **Proof.** By definition of  $\iota_k$  and  $Q(\eta, \iota_k)$ , we have

$$Q(\eta, \iota_k) = \begin{cases} 0 & \text{if } \eta > k + \frac{1}{2\psi} \\ \frac{1}{2} - \psi(\eta - k) & \text{if } \eta \in \left[k - \frac{1}{2\psi}, k + \frac{1}{2\psi}\right] \\ 1 & \text{if } \eta < k - \frac{1}{2\psi} \end{cases}$$

By Assumption 1, there is a unique value  $\eta \in \left[k - \frac{1}{2\psi}, k + \frac{1}{2\psi}\right]$  such that  $Q(\eta, \iota_k) = \frac{1-\tau}{\sigma} - \delta\ell + \eta$ . Define function  $k \mapsto \varphi(k) \in \mathbb{R}$  as the mapping that associates to every k, the amount f that solves  $Q(k + f, \iota_k) = \frac{1-\tau}{\sigma} - \delta\ell + k + f$ . Intuitively,  $\varphi(k)$  represents the increase (or decrease) with respect to the threshold k necessary to make the fraction of revolters exactly equal to the one needed to overthrow the regime. Then, if all activists follow cutoff strategy  $\iota_k$ , the revolt succeeds in overthrowing Player H if and only if  $\eta \in [\underline{\eta}, k + \varphi(k)]$ . As a result,  $w(k, \iota_k) = \int_{k-\frac{1}{2\psi}}^{k+\varphi(k)} gd\varepsilon - \int_{k+\varphi(k)}^{k+\frac{1}{2\psi}} \pi d\varepsilon$ . By construction,  $w(k, \iota_k)$  is a continuous function. Since the solution to equation  $Q(x, \iota_k) = \frac{1-\tau}{\sigma} - \delta\ell + k + (1+\psi)\varphi(k)$ . Totally differentiating the previous expression, we obtain  $\varphi'(k) = -\frac{1}{1+\psi} < 0$ . Thus  $w(k, \alpha_k)$  is a continuous and strictly decreasing function of k.

The meaning of Lemma 2 is straightforward: if activists follow cutoff strategy  $\iota_k$ , the utility of the "marginal revolter" (namely, the revolter who receives the cutoff signal k) is decreasing in the level of the cutoff. Indeed, if the cutoff is high, less people revolt and the utility of revolting is lower as well.

**Lemma 3** There is a unique  $\varepsilon^*$  such that in any equilibrium of the revolt subgame, an activist revolts if and only if  $\varepsilon_i < \varepsilon^*$ .

**Proof.** First, notice that there is a unique  $\varepsilon^*$  such that  $w(\varepsilon^*, \iota_{\varepsilon^*}) = 0$ . Indeed, by the previous lemma,  $w(k, \iota_k)$  is continuous and strictly decreasing in k. If  $k < \underline{\eta} - \frac{1}{2\psi}$ , each activist knows that by revolting Player H will be overthrown. Thus,  $w(k, \iota_k) = g > 0$ . Instead, if  $k > \overline{\eta} + \frac{1}{2\psi}$ , the activist knows that the revolt fails even if all activists revolt. Thus  $w(k, \iota_k) = -\pi < 0$ . As a result, we can find a unique  $\varepsilon^*$  such that  $w(\varepsilon^*, \iota_{\varepsilon^*}) = 0$ . Now, pick any equilibrium of the revolt subgame and let  $\alpha(\cdot)$  describe the fraction of activists who revolt in such equilibrium. Define:

$$\underline{\varepsilon} = \inf \{ \varepsilon_i \mid \alpha(\varepsilon_i) < 1 \},\$$
  
$$\overline{\varepsilon} = \sup \{ \varepsilon_i \mid \alpha(\varepsilon_i) > 0 \},\$$

By definition,  $\overline{\varepsilon} \geq \sup \{\varepsilon_i \mid \alpha(\varepsilon_i) \in (0,1)\} \geq \inf \{\varepsilon_i \mid \alpha(\varepsilon_i) \in (0,1)\} \geq \underline{\varepsilon}$ . If  $\alpha(\varepsilon_i) < 1$ , activists must weakly prefer not revolting than revolting. By continuity, the same holds at  $\underline{\varepsilon}$ . Thus,  $w(\underline{\varepsilon}, \alpha) \leq 0$ . Consider  $w(\underline{\varepsilon}, \iota_{\underline{\varepsilon}})$ . By definition of  $\underline{\varepsilon}, \iota_{\underline{\varepsilon}} < \alpha$ . Lemma 1 thus implies  $w(\underline{\varepsilon}, \iota_{\underline{\varepsilon}}) \leq w(\underline{\varepsilon}, \alpha) \leq 0$ . Since we know that  $w(\varepsilon^*, \iota_{\varepsilon^*}) = 0$ , Lemma 2 implies that  $\underline{\varepsilon} \geq \varepsilon^*$ . A symmetric argument implies that  $\overline{\varepsilon} \leq \varepsilon^*$ . Thus  $\underline{\varepsilon} \geq \varepsilon^* \geq \overline{\varepsilon}$ . We conclude that  $\underline{\varepsilon} = \overline{\varepsilon} = \varepsilon^*$ . Thus,  $\alpha(\cdot)$  is equal to the cutoff strategy with cutoff  $\varepsilon^*$ .

Lemma 3 implies that the fraction of activists who revolt in state  $\eta$  is given by:

$$Q\left(\eta,\iota_{\varepsilon^*}\right) = \begin{cases} 0 & \text{if } \eta > \varepsilon^* + \frac{1}{2\psi} \\ \frac{1}{2} - \psi\left(\eta - \varepsilon^*\right) & \text{if } \eta \in \left[\varepsilon^* - \frac{1}{2\psi}, \varepsilon^* + \frac{1}{2\psi}\right] \\ 1 & \text{if } \eta < \varepsilon^* - \frac{1}{2\psi} \end{cases}$$

Obviously,  $Q(\eta, \iota_{\varepsilon^*})$  is decreasing in  $\eta$  and strictly decreasing in the interval  $\left[\varepsilon^* - \frac{1}{2\psi}, \varepsilon^* + \frac{1}{2\psi}\right]$ . Instead,  $\frac{1-\tau}{\sigma} - \delta \ell + \eta$  is strictly increasing in  $\eta$ . Thus, the two curves cross only once. Let  $\eta^*$  be the realization of  $\eta$  at which this happens.

By definition, in equilibrium if an activist receives signal  $\varepsilon^*$ , he is indifferent between revolting or not. Thus, the following equation must hold:

$$\int_{\varepsilon^* - \frac{1}{2\psi}}^{\eta^*} \psi g d\eta - \int_{\eta^*}^{\varepsilon^* + \frac{1}{2\psi}} \psi \pi d\eta = 0.$$
(13)

Moreover, Player H survives if and only if the realization of  $\eta$  is below the value  $\eta^*$  that satisfies the following equation:

$$\int_{\eta^* - \frac{1}{2\psi}}^{\varepsilon^*} \psi dz = \frac{1 - \tau}{\sigma} - \delta\ell + \eta^*.$$
(14)

Equation 14 says that when the realization of  $\eta$  is  $\eta^*$ , the fraction of individuals who revolt (i.e., the fraction of individuals who observe a signal below  $\varepsilon^*$ ) is exactly equal to the fraction needed to overthrow the regime.

Equations (13) and (14) define a system of two equations in two unknowns ( $\varepsilon^*$  and  $\eta^*$ ). The solution of such system is

$$\eta^* = \frac{g}{g+\pi} - \frac{1-\tau}{\sigma} + \delta\ell,$$
  
$$\varepsilon^* = \frac{g-\pi}{2\psi \left(g+\pi\right)} + \frac{g}{g+\pi} - \frac{1-\tau}{\sigma} + \delta\ell$$

The expression of  $S(\sigma, \tau, \pi, \delta)$  follows from the fact that  $\eta$  is uniformly distributed in the

interval  $[\eta, \overline{\eta}]$ .

#### Proof of Proposition 2 and the Optimal Policy Vector.

Define  $\bar{y}(r) := \sqrt{g} - g + B(1+r)$ . In words,  $\bar{y}(r)$  is the cutoff level on citizens' income that determines whether the constraint  $\tau \leq 1$  binds or not. Let  $\pi_{|\delta=1}^*$  and  $\tau_{|\delta=1}^*$ (resp.,  $\pi_{|\delta=0}^*$  and  $\tau_{|\delta=0}^*$ ) be the optimal level of revolt prevention and taxation conditional on Player *H* choosing to default (resp., not to default). Formally, these values solve (6) if we set  $\delta = 1$  (resp.,  $\delta = 0$ ). In line with the discussion in the main text

$$\pi^*_{|\delta=1} = \sqrt{g} - g \ge \pi^*_{|\delta=0} = \begin{cases} \sqrt{g} - g & \text{if } \frac{1}{\sigma} \ge \bar{y}\left(r\right) \\ \max\left\{0, \frac{1}{\sigma} - B\left(1+r\right)\right\} & \text{if } \frac{1}{\sigma} < \bar{y}\left(r\right) \end{cases}$$

and

$$\tau_{|\delta=0}^{*} = \begin{cases} \sigma \left[\sqrt{g} - g + B\left(1 + r\right)\right] & \text{if } \frac{1}{\sigma} \ge \bar{y}\left(r\right) \\ 1 & \text{if } \frac{1}{\sigma} < \bar{y}\left(r\right) \end{cases} > \tau_{|\delta=1}^{*} = \sigma \left[\sqrt{g} - g\right]$$

Thus, a default leads to a (weak) increase in the amount invested in revolt prevention and to a decrease in the level of taxation.

Let  $\Delta(\sigma, \ell \mid r)$  be Player *H*'s benefit from defaulting when the level of sanctions is  $\sigma$ , the cost of defaulting is  $\ell$  and the level of interest rate is equal to *r*. Formally,  $\Delta(\sigma, \ell \mid r) = \upsilon \left[ S\left(\sigma, \tau^*_{|\delta=1}, \pi^*_{|\delta=1}, 1\right) - S\left(\sigma, \tau^*_{|\delta=0}, \pi^*_{|\delta=0}, 0\right) \right]$ . Exploiting Proposition 1, we get:

$$\Delta\left(\sigma,\ell\mid r\right) = \frac{\upsilon}{\overline{\eta}-\underline{\eta}} \cdot \left[\frac{\tau_{|\delta=0}^{*} - \tau_{|\delta=1}^{*}}{\sigma} + \frac{g \cdot \left(\pi_{|\delta=1}^{*} - \pi_{|\delta=0}^{*}\right)}{\left(g + \pi_{|\delta=1}^{*}\right)\left(g + \pi_{|\delta=0}^{*}\right)} - \ell\right].$$
 (15)

By looking at the squared bracket in (15), we can identify the three separate forces we discuss in the main text. First, a default enables Player H to reduce taxation. Second, a default enables Player H to increase the investment in revolt prevention (notice that this second channel is at play only if  $\pi_{|\delta=1}^* > \pi_{|\delta=0}^*$ , namely only if  $\tau \leq 1$  is binding). Third, a default generates a disutility for the citizens equal to  $\ell$ .

Substituting for  $\left(\pi^*_{|\delta=1}, \pi^*_{|\delta=0}, \tau^*_{|\delta=1}, \tau^*_{|\delta=0}\right)$  in (15), we can further conclude that

$$\Delta\left(\sigma,\ell\mid r\right) = \begin{cases} \frac{\upsilon}{\overline{\eta}-\underline{\eta}} \cdot \left[B\left(1+r\right)-\ell\right] & \text{if } \frac{1}{\sigma} \ge \bar{y}\left(r\right) \\ \frac{\upsilon}{\overline{\eta}-\underline{\eta}} \cdot \left[\frac{g}{g+\frac{1}{\sigma}-B(1+r)} - 2\sqrt{g} + \frac{1}{\sigma} + g - \ell\right] & \text{if } \frac{1}{\sigma} < \bar{y}\left(r\right) \end{cases}$$

Obviously,  $\Delta(\sigma, \ell \mid r)$  is decreasing in  $\ell$ . Moreover, observe that (i)  $\Delta(\sigma, \ell \mid r)$  is piecewise increasing in r, (ii)  $\bar{y}(r)$  is increasing in r, and (iii)  $\Delta(\sigma, \ell \mid r)$  is higher if  $\frac{1}{\sigma} < \bar{y}(r)$  than

if  $\frac{1}{\sigma} \geq \bar{y}(r)$ .<sup>27</sup> Then,  $\Delta(\sigma, \ell \mid r)$  is increasing in r.

The optimal policy vector for Player H is described in the following proposition

**Lemma 4** Suppose the debt is priced and Assumptions 1 and 2 hold. Then, the optimal policy vector  $(T(\sigma, \ell \mid r), P(\sigma, \ell \mid r), D(\sigma, \ell \mid r))$  is given by:<sup>28</sup>

$$P(\sigma, \ell \mid r) = \begin{cases} \max\left\{0, \frac{1}{\sigma} - B(1+r)\right\} & \text{if } \frac{1}{\sigma} < \bar{y}(r) \text{ and } \Delta(\sigma, \ell \mid r) \le 0; \\ \sqrt{g} - g & \text{otherwise.} \end{cases}$$
(16)

$$D(\sigma, \ell \mid r) = \begin{cases} 0 & \text{if } \Delta(\sigma, \ell \mid r) \le 0; \\ 1 & \text{if } \Delta(\sigma, \ell \mid r) > 0. \end{cases}$$
(17)

$$T(\sigma, \ell \mid r) = \sigma \left[ P(\sigma, \ell \mid r) + B(1+r)(1 - D(\sigma, \ell \mid r)) \right]$$
(18)

**Proof.** Recall that Player H faces program (6). Substituting for  $S(\sigma, \pi, \delta, \tau)$  in the objective function and exploiting the budget constraint, we can immediately conclude that the objective function is strictly concave in  $\pi$ . Indeed, the first order condition is given by  $\frac{v}{\overline{\eta}-\underline{\eta}}\left(\frac{g}{(g+\pi)^2}-1\right)$ , while the second order condition is equal to  $-\frac{2v}{\overline{\eta}-\underline{\eta}}\frac{g}{(g+\pi)^3} < 0$ . Thus, if we ignore the fact that  $\tau \leq 1$ , we get  $\pi^* = \sqrt{g} - g$ .

If Player H defaults  $(\delta = 1)$ , level  $\sqrt{g} - g$  of overthrow prevention can always be attained. Moreover, since taxation does not entail any benefit per se, it will be set equal to the lowest value that finances  $\sqrt{g} - g$ . As a result,  $\pi_{|\delta=1}^* = \sqrt{g} - g$  and  $\tau_{|\delta=1}^* = \sigma \pi_{|\delta=1}^*$ . Now, suppose that Player H does not default  $(\delta = 0)$ . Then, level of overthrow prevention  $\sqrt{g} - g$  can be attained if and only if  $1/\sigma \geq \bar{y}(r)$ . In this case,  $\pi_{|\delta=0}^* = \sqrt{g} - g$  and the tax rate will be set residually in order to finance both  $\sqrt{g} - g$  and the service of public debt,  $\tau_{|\delta=0}^* = \sigma \left[\sqrt{g} - g + B(1+r)\right]$ . Instead, if  $1/\sigma < \bar{y}(r)$ , the first-best investment in revolt prevention cannot be attained. Thus, (6) is solved by choosing the highest possible tax rate and by setting the investment in revolt prevention residually. Formally,  $\tau_{|\delta=0}^* = 1$  and  $\pi_{|\delta=0}^* = \max \left\{ 0, \frac{1}{\sigma} - B(1+r) \right\}$ . Obviously, Player H defaults if and only if the benefit from doing so,  $\Delta(\sigma, \ell \mid r)$ , is positive. Furthermore, the optimal level of revolt prevention can be derived noticing that the first-best level is attainable if either Player H defaults or she does not default, but the income level of voters is sufficiently high to finance  $\pi^*$ . Finally, the optimal tax rate is obtained by (3).

Then, a political default occurs if  $\Delta(\sigma, \ell \mid r) \ge 0$ . As a result, the probability of a political default is equal to  $F(\sigma \mid r) = \Pr \{\Delta(\sigma, \ell \mid r) \ge 0\}$ . Since  $\Delta(\sigma, \ell \mid r)$  is de-

<sup>&</sup>lt;sup>27</sup>To see this last result, observe that if  $\frac{1}{\sigma} \leq \bar{y}(r)$ , the differences  $\tau^*_{|\delta=0} - \tau^*_{|\delta=1}$  and  $\pi^*_{|\delta=1} - \pi^*_{|\delta=0}$  are both larger than in the case in which  $\frac{1}{\sigma} > \bar{y}(r)$ .

 $<sup>^{28}</sup>$ In Lemma 4, we assume that, whenever indifferent between defaulting or not, Player H does not default. None of our results hinges on this tie-breaking rule.

creasing in  $\ell$ , we can have two cases (it is immediate to verify that  $\Delta(\sigma, 0 | r) > 0$ ). If  $\Delta(\sigma, \ell_H | r) \ge 0$ , then defaulting is always profitable and, consequently, a political default happens with probability 1. As discussed in Footnote 13, we rule out this case assuming that  $\ell_H$  is sufficiently high. Instead, if  $\Delta(\sigma, \ell_H | r) < 0$ , a political default takes place (resp., does not take place) if  $\ell$  is sufficiently low (resp., high). In this last case, exploiting the distributional assumption on  $\ell$ , we can conclude that  $F(\sigma | r) = \ell^*$ , where  $\ell^*$  is the unique solution  $\ell$  of  $\Delta(\sigma, \ell | r) = 0$ . As a result,

$$F\left(\sigma \mid r\right) = \begin{cases} \frac{1}{\ell_H} B\left(1+r\right) & \text{if } \frac{1}{\sigma} \ge \bar{y}\left(r\right) \\ \frac{1}{\ell_H} \left[\frac{1}{\sigma} + g\left(\frac{1}{g+\frac{1}{\sigma} - B(1+r)} + 1\right) - 2\sqrt{g}\right] & \text{if } \frac{1}{\sigma} < \bar{y}\left(r\right) \end{cases}$$

We can immediately conclude that  $\partial F(\sigma \mid r) / \partial r > 0$  and  $\partial F(\sigma \mid r) / \partial \sigma \ge 0$ . To see why, observe that if  $1/\sigma \ge \bar{y}(r)$ , then  $F(\sigma \mid r)$  is constant in  $\sigma$ , while if  $1/\sigma < \bar{y}(r)$ , then  $\frac{\partial F(\sigma \mid r)}{\partial \sigma} = -\frac{1}{\sigma^2} + g\left(\frac{\frac{1}{\sigma^2}}{\left(g + \frac{1}{\sigma} - B(1+r)\right)^2}\right) > -\frac{1}{\sigma^2} + \frac{1}{\sigma^2} = 0$ , where the inequality follows from observing that, since  $1/\sigma < \bar{y}(r)$ , we have  $0 < g + 1/\sigma - B(1+r) < \sqrt{g}$ .<sup>29</sup>

#### **Proof of Proposition 3.**

By definition,  $R(\sigma)$  is the lowest root r of equation  $1 + \tilde{r} = (1 + r)(1 - F(\sigma | r))$ . Suppose first that  $\frac{1}{\sigma} \ge \sqrt{g} - g + B(1 + R(\sigma))$ . In this case,  $F(\sigma | R(\sigma)) = \frac{B(1+R(\sigma))}{\ell_H}$ . Thus, the no-arbitrage condition yields a solution if and only if  $B \le \ell_H / [4(1 + \tilde{r})]$ , in which case

$$R(\sigma) = \ell_H \left(\frac{1 - \sqrt{1 - \frac{4B(1+\tilde{r})}{\ell_H}}}{2B}\right) - 1.$$
(19)

Substituting for  $R(\sigma)$ , inequality  $\frac{1}{\sigma} \ge \pi^* + B(1 + R(\sigma))$  becomes  $\sigma \le 2/[2(\sqrt{g} - g) + \ell_H(1 - \sqrt{(1 - 4B(1 + \tilde{r}))/\ell_H})] := \hat{\sigma}$ . Thus, if  $\sigma \le \hat{\sigma}$  and  $B \le \ell_H/[4(1 + \tilde{r})]$ , the debt is priced and the interest rate is constant and equal to (19). Instead, if  $\sigma \le \hat{\sigma}$  and  $B > \ell_H/[4(1 + \tilde{r})]$ , the debt is not priced.

Now suppose that  $\frac{1}{\sigma} < \sqrt{g} - g + B(1 + R(\sigma))$ . Then, the equilibrium interest rate is given by the lowest value of r that solves

$$(1+r)\left(1-\frac{\frac{1}{\sigma}+\frac{g}{g+\frac{1}{\sigma}-B(1+r)}+g-2\sqrt{g}}{\ell_H}\right)-(1+\tilde{r})=0.$$
(20)

It is easy to show that if  $\sigma \to \hat{\sigma}$ , the interest rate coincides with (19). Let  $Q(\sigma, r)$ 

 $<sup>2^{9}</sup>g + 1/\sigma - B(1+r) > 0$  holds as we are assuming that the debt is priced, thus  $1/\sigma \ge B(1+r)$ . If this inequality were not satisfied, the country would have no possibility to repay the outstanding debt and an automatic default would occur.

be the left-hand side of (20). By the implicit function theorem, (20) defines r as a function of  $\sigma$ ,  $R(\sigma)$ . Moreover,  $\frac{dR(\sigma)}{d\sigma} = -\frac{\partial Q(\sigma,r)}{\partial \sigma} / \frac{\partial Q(\sigma,r)}{\partial r}$ .  $\frac{\partial Q(\sigma,r)}{\partial \sigma}$  has the same sign of  $(1 - 1/[g + 1/\sigma - B(1 + r)])$ . Since we are considering the case in which  $\frac{1}{\sigma} < \sqrt{g} - g + B(1 + R(\sigma))$ , we have that  $(1 - 1/[g + 1/\sigma - B(1 + r)]) < 1 - 1/g < 0$ . Thus,  $\frac{\partial Q(\sigma,r)}{\partial \sigma} < 0$ . Thus,  $\frac{dR(\sigma)}{d\sigma}$  is positive if and only if  $\frac{\partial Q(\sigma,r)}{\partial r} > 0$ , which holds for the lowest root of (20) as (20) has two roots and  $Q(\sigma, \tilde{r}) < 0$ .

Now we show that the probability of a political default,  $F(\sigma | R(\sigma))$ , is weakly increasing in  $\sigma$ . If  $\sigma \leq \hat{\sigma}$ ,  $R(\sigma)$  is constant in  $\sigma$ . Furthermore, as we have shown in the proof of Proposition 2,  $F(\sigma | R(\sigma)) = B(1 + R(\sigma))/\ell_H$ . Then it is immediate to see that, in this case,  $F(\sigma | R(\sigma))$  is constant in  $\sigma$ . Now suppose that  $\sigma > \hat{\sigma}$  (so that,  $\frac{1}{\sigma} \leq \sqrt{g} - g + B(1 + R(\sigma))$ ). Then,  $R(\sigma)$  is increasing in  $\sigma$  and

$$F\left(\sigma \mid R\left(\sigma\right)\right) = \frac{1}{\ell_{H}} \left[\frac{1}{\sigma} + g\left(\frac{1}{g + \frac{1}{\sigma} - B\left(1 + R\left(\sigma\right)\right)} + 1\right) - 2\sqrt{g}\right]$$

As a result,

$$\frac{dF\left(\sigma \mid R\left(\sigma\right)\right)}{d\sigma} = \frac{1}{\ell_{H}} \left[ -\frac{1}{\sigma^{2}} + g\left(\frac{\frac{1}{\sigma^{2}} + B\frac{dR(\sigma)}{d\sigma}}{\left(g + \frac{1}{\sigma} - B\left(1 + R\left(\sigma\right)\right)\right)^{2}}\right) \right]$$

Since  $B\frac{dR(\sigma)}{d\sigma} \ge 0$ , we can reason as at the end of the proof of Proposition 2 and conclude that  $\frac{dF(\sigma|R(\sigma))}{d\sigma} > 0$ . Moreover,  $\lim_{\sigma \to \hat{\sigma}^+} R(\sigma) = \frac{\ell_H}{2B} \left(1 - \sqrt{1 - \frac{4B(1+\tilde{r})}{\ell_H}}\right) - 1$ . We conclude that  $F(\sigma \mid R(\sigma))$  is weakly increasing in  $\sigma$ .

### **Proof of Proposition 4.**

Exploiting equations (1), (3) and (5), we can rewrite the expected survival probability as:

$$\hat{S}(\sigma) = \int_{0}^{\ell_{H}} \left[ 1 - \frac{1}{\overline{\eta} - \underline{\eta}} \left( \frac{g}{g + P(\sigma, \ell \mid R(\sigma))} - \frac{1}{\sigma} + P(\sigma, \ell \mid R(\sigma)) + (1 + R(\sigma))(1 - D(\sigma, \ell \mid R(\sigma))) + \delta \ell \right) \right] \frac{d\ell}{\ell_{H}}$$

By definition,  $\int_{0}^{\ell_{H}} (1 + R(\sigma)) (1 - D(\sigma, \ell \mid R(\sigma))) \frac{d\ell}{\ell_{H}} = (1 + R(\sigma)) (1 - F(\sigma \mid R(\sigma))).$ 

Furthermore, by (8), we know that  $d(1 + R(\sigma))(1 - F(\sigma \mid R(\sigma)))/d\sigma = 0$ . Therefore

$$\begin{aligned} \frac{d\hat{S}\left(\sigma\right)}{d\sigma} &= \frac{d}{d\sigma} \left( \int_{0}^{\ell_{H}} \left[ 1 - \frac{1}{\overline{\eta} - \underline{\eta}} \left( \frac{g}{g + P\left(\sigma, \ell \mid R\left(\sigma\right)\right)} - \frac{1}{\sigma} + P\left(\sigma, \ell \mid R\left(\sigma\right)\right) + \delta\ell \right) \right] \frac{d\ell}{\ell_{H}} \right) = \\ &= \int_{0}^{\ell_{H}} \frac{1}{\overline{\eta} - \underline{\eta}} \left[ -\frac{1}{\sigma^{2}} - \left( 1 - \frac{g}{\left(g + P\left(\sigma, \ell \mid R\left(\sigma\right)\right)\right)^{2}} \right) \frac{\partial P\left(\sigma, \ell \mid R\left(\sigma\right)\right)}{\partial\sigma} \right] \frac{d\ell}{\ell_{H}} \end{aligned}$$

Obviously, if  $\frac{1}{\sigma} \geq \bar{y}(r)$ , then  $\frac{\partial P(\sigma, \ell | R(\sigma))}{\partial \sigma} = 0$  and  $\frac{d\hat{S}(\sigma)}{d\sigma} = -\frac{1}{\sigma^2(\bar{\eta}-\underline{\eta})} < 0$ . Instead, if  $\frac{1}{\sigma} < \bar{y}(r)$ , then

$$\frac{d\hat{S}\left(\sigma\right)}{d\sigma} = -\frac{1}{\sigma^{2}\left(\overline{\eta} - \underline{\eta}\right)} - \int_{\ell^{*}}^{\ell_{H}} \frac{1}{\overline{\eta} - \underline{\eta}} \left(1 - \frac{g}{\left(g + P\left(\sigma, \ell \mid R\left(\sigma\right)\right)\right)^{2}}\right) \frac{\partial P\left(\sigma, \ell \mid R\left(\sigma\right)\right)}{\partial \sigma} \frac{d\ell}{\ell_{H}}$$

Moreover, if  $\ell < \ell^*$ , then  $P(\sigma, \ell \mid R(\sigma)) = \sqrt{g} - g$  and  $\frac{\partial P(\sigma, \ell \mid R(\sigma))}{\partial \sigma} = 0$ , while if  $\ell > \ell^*$ , then  $P(\sigma, \ell \mid R(\sigma)) = \frac{1}{\sigma} - (1 + R(\sigma)) B$  and  $\frac{\partial P(\sigma, \ell \mid R(\sigma))}{\partial \sigma} = -\frac{1}{\sigma^2} - B\left(1 + \frac{dR(\sigma)}{d\sigma}\right) < 0$ . In the former case, we can replicate the analysis carried out for the case  $\frac{1}{\sigma} \ge \bar{y}(r)$  and conclude that  $\frac{d\hat{S}(\sigma)}{d\sigma} < 0$ . In the latter case,  $0 < P(\sigma, \ell \mid R(\sigma)) \le \sqrt{g} - g$ . Therefore  $\frac{g}{(g + P(\sigma, \ell \mid R(\sigma)))^2} \ge \frac{g}{(g + \sqrt{g} - g)^2} = 1$ . Thus,  $\frac{d\hat{S}(\sigma)}{d\sigma} < 0$  even if  $\frac{1}{\sigma} < \bar{y}(r)$ .

#### **Proof of Proposition 5.**

To prove the statement of the proposition, first notice that for any profile of political weight  $\phi = (\phi_1, ..., \phi_n)$ , maximizing  $\tilde{W}(\sigma \mid b, \phi)$  is equivalent to maximize (10) for a country j whose cost  $c_j$  and level of debt  $b_j$  are respectively given by  $\bar{c}_{\phi}$  and  $\bar{b}_{\phi}$ . Obviously, if  $\phi = \{\frac{1}{n}, \frac{1}{n}, ..., \frac{1}{n}\}$ , our argument still holds and  $\bar{c}_{\phi} = \bar{c}$  and  $\bar{b}_{\phi} = \bar{b}$ . Statement i) in the proposition would then follow if we show that the optimal level of sanctions for a country is decreasing in  $\bar{c}_{\phi}$ . Suppose not. Then, we can find two profiles of political powers  $\bar{c}'_{\phi}$ and  $\bar{c}''_{\phi}$ , with  $\bar{c}'_{\phi} < \bar{c}''_{\phi}$  and two levels of sanctions  $\sigma'$  and  $\sigma''$  with  $\sigma' < \sigma''$  such that:

$$\sigma' \in \arg \max_{\sigma \in [1,\bar{\sigma}]} G(\sigma) + k \cdot \left(1 - \hat{S}(\sigma)\right) - \vec{c}_{\phi}' \cdot F(\sigma \mid R(\sigma)) + b_{\phi} \cdot (1 + \tilde{r})$$
(21)

$$\sigma'' \in \arg \max_{\sigma \in [1,\bar{\sigma}]} G(\sigma) + k \cdot \left(1 - \hat{S}(\sigma)\right) - \vec{c}_{\phi}'' \cdot F(\sigma \mid R(\sigma)) + b_{\phi} \cdot (1 + \tilde{r})$$
(22)

(the actual value of  $b_{\phi}$  plays no role in our argument as it does not effect the maximization).

Since  $\sigma'$  is a maximum when the weighted cost is given by  $\vec{c}_{\phi}$ , we know that  $G(\sigma') + k \cdot \left(1 - \hat{S}(\sigma')\right) - \vec{c}_{\phi} \cdot F(\sigma' \mid R(\sigma'')) \ge G(\sigma'') + k \cdot \left(1 - \hat{S}(\sigma'')\right) - \vec{c}_{\phi} \cdot F(\sigma'' \mid R(\sigma''))$  or equivalently

$$k \cdot \left[\hat{S}\left(\sigma''\right) - \hat{S}\left(\sigma'\right)\right] - \left[G\left(\sigma''\right) - G\left(\sigma'\right)\right] - \vec{c}_{\phi} \cdot \left[F\left(\sigma' \mid R\left(\sigma'\right)\right) - F\left(\sigma'' \mid R\left(\sigma''\right)\right)\right] \ge 0.$$
(23)

Similarly, since  $\sigma''$  is a maximum when the weighted cost is given by  $c''_{\phi}$ , we have:

$$0 \ge k \cdot \left[\hat{S}\left(\sigma''\right) - \hat{S}\left(\sigma'\right)\right] - \left[G\left(\sigma''\right) - G\left(\sigma'\right)\right] - \bar{c}_{\phi}'' \cdot \left[F\left(\sigma' \mid R\left(\sigma'\right)\right) - F\left(\sigma'' \mid R\left(\sigma''\right)\right)\right].$$
(24)

Since  $F(\sigma | R(\sigma))$  is weakly increasing in  $\sigma$ , we can have one of two possible cases: (i)  $F(\sigma' | R(\sigma')) = F(\sigma'' | R(\sigma''))$ , or (ii)  $F(\sigma' | R(\sigma')) < F(\sigma'' | R(\sigma''))$ . In the former case, because  $\hat{S}(\sigma)$  is weakly decreasing in  $\sigma$  and  $G(\sigma)$  is strictly increasing in  $\sigma$ ,  $k \cdot \hat{S}(\sigma') - G(\sigma') - \vec{c}_{\phi}' \cdot F(\sigma' | R(\sigma')) < k \cdot \hat{S}(\sigma'') - G(\sigma'') - \vec{c}_{\phi}' \cdot F(\sigma'' | R(\sigma''))$ , which contradicts (24). In the latter case,  $F(\sigma' | R(\sigma')) < F(\sigma'' | R(\sigma''))$  and then

$$0 \leq k \cdot \left[ \hat{S}(\sigma'') - \hat{S}(\sigma') \right] - \left[ G(\sigma'') - G(\sigma') \right] - \vec{c}'_{\phi} \cdot \left[ F(\sigma' \mid R(\sigma')) - F(\sigma'' \mid R(\sigma'')) \right] < k \cdot \left[ \hat{S}(\sigma'') - \hat{S}(\sigma') \right] - \left[ G(\sigma'') - G(\sigma') \right] - \vec{c}''_{\phi} \cdot \left[ F(\sigma' \mid R(\sigma')) - F(\sigma'' \mid R(\sigma'')) \right]$$

contradicting again (24). Statement ii) follows applying the same steps, once we substitute in the previous expression for  $\bar{c}_{\phi} = a + \bar{b}_{\phi}$ .

## 8.2 Additional Empirical Evidence

## **TABLES A1-A9 HERE**

## References

- Allen, S. H. (2008). The domestic political costs of economic sanctions. Journal of Conflict Resolution, 52(6), 916-944.
- [2] Allen, S. H., & Lektzian, D. J. (2013). Economic sanctions: A blunt instrument?. Journal of Peace Research, 50(1), 121-135.
- [3] Amador, M. (2003). A political economy model of sovereign debt repayment. *Manuscript*, Stanford University.
- [4] Anderson, R. W., & Sundaresan, S. (1996). Design and valuation of debt contracts. *Review of financial studies*, 9(1), 37-68.
- [5] Andreasen E., Sandleris, G., & Van der Ghote, A. (2016). The Political Economy of Sovereign Defaults. *Manuscript*, Princeton University.
- [6] Angeletos, G. M., Hellwig, C., & Pavan, A. (2007). Dynamic global games of regime change: Learning, multiplicity, and the timing of attacks. *Econometrica*, 75(3), 711-756.
- [7] Arellano, C. (2008). Default risk and income fluctuations in emerging economies. The American Economic Review, 98(3), 690-712.
- [8] Atkeson, A., 2000. Discussion of Morris and Shin's Rethinking Multiple Equilibria in Macroeconomic Modelling. *Research Department Staff Report*, Federal Reserve Bank of Minneapolis.
- [9] Baldwin, D. (2000). The sanctions debate and the logic of choice. International Security, 24(3), 80–107.
- [10] Borensztein, E., & Panizza, U. (2009). The costs of sovereign default. IMF Economic Review, 56(4), 683-741.
- [11] Brewer, T. L., & Rivoli, P. (1990). Politics and perceived country creditworthiness in international banking. *Journal of Money, Credit and Banking*, 22(3), 357-369.
- [12] Bulow, J. I., & Rogoff, K. S. (1989a). Sovereign debt: Is to forgive to forget? American Economic Review, 79(1), 43-50.
- [13] Bulow, J., & Rogoff, K. (1989b). A constant recontracting model of sovereign debt. Journal of Political Economy, 97(1), 155-178.

- [14] Carlsson, H. and Van Damme, E., 1993. "Global games and equilibrium selection." Econometrica, 61(5): 989-1018.
- [15] Cole, H. L., & Kehoe, P. J. (1998). Models of sovereign debt: Partial versus general reputations. *International Economic Review*, 39(1), 55-70.
- [16] Cruces, J., & Trebesch, C. (2013). Sovereign Defaults: The Price of Haircuts. American Economic Journal: Macroeconomics, 5(3), 85-117.
- [17] Cuadra, G., & Sapriza, H. (2008). Sovereign default, interest rates and political uncertainty in emerging markets. *Journal of international Economics*, 76(1), 78-88.
- [18] De Mesquita, B. B., & Smith, A. (2009). Political survival and endogenous institutional change. *Comparative Political Studies*, 42(2), 167-197.
- [19] De Mesquita, B. B., & Smith, A. (2010). Leader survival, revolutions, and the nature of government finance. *American Journal of Political Science*, 54(4), 936-950.
- [20] Drezner, D. W. (1999). The sanctions paradox: Economic statecraft and international relations, Cambridge University Press.
- [21] Drezner, D. & Drury, A. (2000). Bargaining, enforcement, and multilateral sanctions: When is cooperation counterproductive?, *International Organization*, 54(1), 73-102.
- [22] Eaton, J., & Fernandez, R. (1995). Sovereign debt. Handbook of international economics, 3, 2031-2077.
- [23] Edmond, C. (2013). Information Manipulation, Coordination, and Regime Change. *Review of Economic Studies*, 80, 1422-1458.
- [24] English, W. B. (1996). Understanding the costs of sovereign default: American state debts in the 1840's. *The American Economic Review*, 86(1), 259-275.
- [25] Escribà-Folch, A. & Wright, J. (2010). Dealing with tyranny: International sanctions and the survival of authoritarian rulers. *International Studies Quarterly*, 54, 335–359.
- [26] Galtung, J. (1967). On the Effects of International Economic Sanctions, With Examples from the Case of Rhodesia. World Politics, 19(3), 378-416.
- [27] Gelos, R. G., Sahay, R., & Sandleris, G. (2011). Sovereign borrowing by developing countries: What determines market access? *Journal of international Economics*, 83(2), 243-254.

- [28] Gennaioli, N., Martin, A., & Rossi, S. (2014). Sovereign default, domestic banks, and financial institutions. *The Journal of Finance*, 69(2), 819-866.
- [29] Grossman, G., Manekin, D. & Margalit Y. (2017). How Sanctions Affect Public Opinion in Target Countries: Experimental Evidence from Israel. Mimeographed.
- [30] Guiso, L., Sapienza, P., & Zingales, L. (2013). The determinants of attitudes toward strategic default on mortgages. *The Journal of Finance*, 68(4), 1473-1515.
- [31] Guembel, A., & Sussman, O. (2009). Sovereign debt without default penalties. The Review of Economic Studies, 76(4), 1297-1320.
- [32] Hufbauer, G. C., Schott, J.J., & Elliott, K. A. (2009). Economic Sanctions Reconsidered, 3rd Edition, Peterson Institute Press: All Books, Peterson Institute for International Economics, number 4129, 03.
- [33] Hultman, L., & Peksen, D. (2015). Successful or Counterproductive Coercion? The Effect of International Sanctions on Conflict Intensity. *Journal of Conflict Resolution*, forthcoming.
- [34] Kaempfer, W. H., & Lowenberg, A. D. (1988). The theory of international economic sanctions: A public choice approach. *The American Economic Review*, 78(4), 786-793.
- [35] Kaempfer, W. H., & Lowenberg, A. D. (1992). Using threshold models to explain international relations. *Public Choice*, 73(4), 419-443.
- [36] Kaempfer, W. H., & Lowenberg, A. D. & Mertens W. (2004). International Economic Sanctions Against a Dictator. *Economics & Politics*, 16(1), 29-51.
- [37] Kohlscheen, E. (2007). Why are there serial defaulters? Evidence from constitutions. The Journal of Law and Economics, 50(4), 713-730.
- [38] Lacy, D. & Niou, E. M. S. (2004). A Theory of Economic Sanctions and Issue Linkage: The Roles of Preferences, Information, and Threats. *Journal of Politics*, 66, 25–42.
- [39] Lektzian, D., & Souva, M. (2007). An institutional theory of sanctions onset and success. Journal of Conflict Resolution, 51(6), 848-871.
- [40] Licht, A. A. (2017). Hazards or Hassles The Effect of Sanctions on Leader Survival. Political Science Research and Methods, 5(01), 143-161.

- [41] Marinov, N., (2005). Do economic sanctions destabilize country leaders? American Journal of Political Science, 49, 564–576.
- [42] Mella-Barral, P., & Perraudin, W. (1997). Strategic debt service. The Journal of Finance, 52(2), 531-556.
- [43] Mitchener, K. J., & Weidenmier, M. D. (2010). Supersanctions and sovereign debt repayment. Journal of International Money and Finance, 29(1), 19-36.
- [44] Morgan, T. C., N. Bapat, and Y. Kobayashi. 2014. "The Threat and Imposition of Sanctions: Updating the TIES dataset." *Conflict Management and Peace Science* 31(5): 541-558.
- [45] Morris, S. & Shin, H. S. (1998). Unique Equilibrium in a Model of Self-Fulfilling Currency Attacks. American Economic Review, 88(3), 587-97.
- [46] Morris, S. and Shin, H.S., 2001. Global games: theory and applications in Advances in Economics and Econometrics (Proceedings of the Eighth World Congress of the Econometric Society), edited by M. Dewatripont, L. Hansen and S. Turnovsky, Cambridge University Press
- [47] Neuenkirch, M., & Neumeier, F. (2015). The impact of UN and US economic sanctions on GDP growth. European Journal of Political Economy, 40, 110-125.
- [48] Oechslin, M. (2014). Targeting autocrats: Economic sanctions and regime change. European Journal of Political Economy, 36, 24-40
- [49] Panizza, U., Sturzenegger, F., & Zettelmeyer, J. (2009). The economics and law of sovereign debt and default. *Journal of Economic Literature*, 47(3), 651-698.
- [50] Pape, R. A. (1997). Why economic sanctions do not work. International security, 22(2), 90-136.
- [51] Passarelli, F. and Tabellini, G. (2017). Emotions and Political Unrest, Journal of Political Economy, vol. 125(3), 903-946.
- [52] Peksen, D. (2009). Better or worse? The effect of economic sanctions on human rights. *Journal of Peace Research*, 46(1), 59-77.
- [53] Peksen, D., & Drury, A. C. (2010). Coercive or corrosive: The negative impact of economic sanctions on democracy. *International Interactions*, 36(3), 240-264.

- [54] Reinhart C. M. and K. S. Rogoff. 2010. Dates for Banking Crisis, Currency Crashes, Sovereign Domestic or External Default (or Restructuring), Inflation Crises, and Stock Market Crashes (Varieties).
- [55] Sandleris, G. (2016). The costs of sovereign default: Theory and empirical evidence. Economía, 16(2), 1-27.
- [56] Toke S. A. & Albornoz F. (2011). Political regimes and foreign intervention. Journal of Development Economics, 94(2), 192-201.
- [57] Tomz, M., & Wright, M. L. (2013). Empirical research on sovereign debt and default. Annual Review of Economics, 5(1), 247-272.
- [58] Van Rijckeghem, C., & Weder, B. (2009). Political institutions and debt crises. Public Choice, 138(3-4), 387-408.
- [59] Weidenmier, M. D. (2005). Gunboats, reputation, and sovereign repayment: lessons from the Southern Confederacy. *Journal of International Economics*, 66(2), 407-422.
- [60] Weiss, T. G. (1997). Political Gain and Civilian Pain: Humanitarian impacts of economic sanctions. Rowman & Littlefield.
- [61] Wilson K.A. (2012). Cross National Time Series Data Archive, User's Manual.
- [62] Wood, R. M. (2008). A Hand upon the Throat of the Nation: Economic Sanctions and State Repression, 1976–2001. International Studies Quarterly, 52(3), 489-513.
- [63] World Development Indicators, The World Bank. 2015.
- [64] Yeyati, E. L., & Panizza, U. (2011). The elusive costs of sovereign defaults. Journal of Development Economics, 94(1), 95-105.

Variable		Eull Comple	Not under	+ +06+	Under	Not in an external	+ +0ct	In an external
A dit tauta		and magnific m.r	sanction	1991-1	sanction	debt crisis	1921-1	debt crisis
Gdp pc	mean	3.905	3.365	* * *	5.332	4.363	* * *	1.175
	s.d.	(6.921)	(6.183)		(8.412)	(7.370)		(1.021)
Gdp pc growth	mean	.068	.069		.065	.074	* * *	.031
	s.d.	(.113)	(.110)		(.118)	(.101)		(.160)
Military	mean	.044	.038	* * *	090.	.044		.045
	s.d.	(.206)	(.192)		(.238)	(.205)		(.208)
Elective	mean	.831	.848	* * *	.787	.844	* * *	.752
	s.d.	(.375)	(.360)		(.410)	(.363)		(.432)
Riot	mean	.247	.229	* * *	.296	.240	* *	.291
	s.d.	(.431)	(.420)		(.457)	(.427)		(.455)
Conflict Index	mean	12.29	11.886	* * *	14.690	11.930	* * *	14.632
	s.d.	(22.051)	(21.301)		(23.794)	(22.328)		(20.179)
N. Observations		3,091	2,243		848	2647		444
Notes: A com	ntrv is de	sfined to he un	der sanction if i	t is targete	d hv at least	t one sanction in a give	n vear. "Mili	tarv" is a dummv

 Table 1. Summary Statistics (full sample)

variable for the country being ruled by a military junta. "Elective" is a dummy for elective institutions. "Riots" is a dummy for at least one riot event. Conflict index is a measure of political unrest computed by the CNTSDA. \*\*\*: difference in means is statistically significant at 1% level in a one-tailed t-test. \*\*: the difference in means is statistically significant at 5% level in a one-tailed t-test.

Variable		Under sanc	tion at tim	e $t - 1$	In exter	nal debt cri	sis at $t-1$
	Not unde	r external	t-test	Under external	Not under	t-test	Under sanction
	debt cr	is at $t$		debt crisis at $t$	sanction at $t$		at $t$
Gdp pc at t+1 me	an    6.0	620	***	1.534	1.053	* * *	1.677
s.d	. (9.0	011)		(1.268)	(779.)		(1.455)
Gdp pc growth at t+1 mee	u	68		.062	.037	* * *	.073
s.d	. (.1	(90		(.165)	(.133)		(.159)
Military	u	61		.039	.054		.031
s.d		(39)		(.194)	(.226)		(.174)
Elective me	an	16	* * *	.619	.821	* * *	.634
s.d		88)		(.487)	(.384)		(.483)
Riot at $t+1$ met	un    .3	00	* *	.219	.294	* * *	.193
s.d	.    .4	(66)		(.415)	(.456)		(.396)
Conflict Index at $t + 1$ mes	nn    14.	869	*	12.112	15.534	* *	11.094
s.d	. (24.	776)		(16.237)	(21.911)		(13.931)
N. Observations	39	57		155	279		161
Notes: Variables are defin t-test. **: the difference in	ed as explaine 1 means is stat	id in Table 1. Listically signi	***: the $d$ : ficant at $5^{\circ}_{1}$	ifference in means is 3 % level in a one-tailec	statistically signifi l t-test. *: the diff	cant at 1% ference in m	level in a one-taille leans is statisticall,

 Table 2. Summary Statistics conditional on sanction and external debt crisis status (full sample)

		Depender	t Variable:	
		Ext. De	ebt Crisis	
	(1)	(2)	(3)	(4)
Sanction in $t-1$	0.0175**	$0.0181^{**}$	$0.0158^{**}$	$0.0352^{*}$
	(0.0085)	(0.0085)	(0.0074)	(0.0192)
Yrs last debt crisis	0.0007	$0.0012^{*}$	0.0012**	0.0013
	(0.0006)	(0.0006)	(0.0006)	(0.0012)
Gdp pc $t-1$		$-0.0016^{**}$	-0.0020**	0.0172
		(0.0007)	(0.0009)	(0.0300)
Gdp pc growth $t-1$		-0.0662	-0.0771	-0.1834
		(0.0596)	(0.0600)	(0.1183)
No inflation crisis $t-2$			-0.0200	-0.0086
			(0.0166)	(0.0292)
Elective			-0.0094	0.0355
			(0.0178)	(0.0548)
Military			0.0337	$0.1309^{*}$
			(0.0451)	(0.0754)
Population			-0.0003***	-0.0009*
			(0.0001)	(0.0005)
Reserves				-0.0006
				(0.0004)
Stock external debt				$0.0022^{**}$
				(0.0011)
Interest ext. payments $t-1$				0.0016
				(0.0077)
Country fixed effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes
Observations	2268	2268	2268	520

 Table 3. Regression of sanctions on external default.

Notes: "Ext. Debt Crisis" is a dummy variable that takes value 1 if the country is in a debt crisis with external creditors. "Sanction in t-1" is a dummy variable that takes value 1 if the country was hit by at least one sanction in the previous year. "No inflation crisis at t-2" is a dummy if the country did not experience a growth in inflation greater or equal than 20% in the last two years. "Reserves" is measured as the % of total reserves over total external debt. "Stock external debt" is the stock of external debt over gross national income. "Interest ext. payments t-1" are the interest payments on external debt in year t-1. Other variables are defined as explained in Table 1. Observations are excluded if the country was under a banking crisis, a domestic debt crisis or and external debt crisis in any of the two previous years. Standard errors (in parenthesis) are clustered at the country level.

	Dependen	t Variable:	Depender	t Variable:
	R	iot	Conflic	ct Index
	(1)	(2)	(3)	(4)
Ext. debt crisis $t-2$	0.0143	0.0391	-199.2276	-284.4170
	(0.0449)	(0.0839)	(297.6479)	(582.0345)
Sanction $t-2$	-0.0124	0.0543	24.3801	193.0626
	(0.0245)	(0.0489)	(185.5962)	(343.3499)
Interaction	-0.1355**	$-0.1738^{**}$	-622.9851*	-961.9269**
	(0.0613)	(0.0676)	(350.9677)	(471.4285)
Gdp pc $t-1$	-0.0027	-0.0429	-48.1140*	73.7383
	(0.0046)	(0.0573)	(27.6019)	(365.6356)
Gdp pc growth $t-1$	-0.1721**	-0.1494	-803.2731**	-1314.7323**
	(0.0688)	(0.1404)	(310.2539)	(604.8249)
Elective	-0.0199	-0.0911	-749.4891**	-810.0128*
	(0.0540)	(0.0901)	(283.4318)	(423.8189)
Military	-0.0052	-0.0785	-209.5711	-477.6651
	(0.0920)	(0.1465)	(344.0653)	(442.0957)
n coups	-0.0082	-0.0296	-198.6667	-222.1251
	(0.0184)	(0.0513)	(126.5693)	(262.5200)
No inflation crisis $t-2$	-0.0333	0.0338	-312.5746**	16.2692
	(0.0357)	(0.0425)	(144.0336)	(180.9811)
Yrs since ext. debt crisis	-0.0015	-0.0018	-13.9061	-22.7088
	(0.0020)	(0.0028)	(12.7632)	(22.8997)
Yrs since executive change	-0.0007	0.0011	-34.1683***	-40.8145***
	(0.0022)	(0.0031)	(9.9019)	(14.1898)
Population	-0.0006	-0.0007	-4.0594	-3.4700
	(0.0005)	(0.0008)	(2.9001)	(5.0884)
Reserves		-0.0006		1.3038
		(0.0010)		(2.5132)
Stock ext. debt		0.0009***		2.5303***
		(0.0002)		(0.7330)
Interest ext. payments		0.0165		92.8730
		(0.0171)		(114.1650)
Country fixed effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes
Observations	2776	978	2775	977

Table 4. Sanctions, External Debt Crisis and Political Unrest

**Notes**: Variables are defined as described in Table 1 and Table 3. The variable "Interaction" is a dummy for the fact that in the last two years the country was both under sanction and in an external debt crisis provided that the former status occurred before the latter. "Yrs since executive change" is the number of years since the last change in the executive power. # coups measures the number of coups. Standard errors (in parenthesis) are clustered at the country level.

Country	Freq. Sanctions	Freq. Ext. Debt Crisis	Country	Freq. Sanctions	Freq. Ext. Debt Crisis
Algeria	.143	.107	Malaysia	.250	0
Argentina	.411	.500	Mauritius	0	0
Australia	.304	0	Mexico	.411	.161
Austria	.054	.054	Morocco	.018	.107
Belgium	.286	0	Myanmar	.232	.071
Bolivia	.125	.304	Netherlands	.286	0
Brazil	.429	.268	New Zeland	.232	0
Canada	.429	0	Nicaragua	.339	.482
Central Africa Republic	.054	.429	Nigeria	.214	.250
Chile	.214	.250	Norway	.054	0
China	.786	0	Panama	.286	.250
Colombia	.125	0	Paraguay	.054	.196
Costa Rica	.161	.232	Peru	.268	.357
Denmark	.054	0	Philippines	.018	.214
Dominican Republic	.250	.250	Poland	.393	.304
Ecuador	.446	.375	Portugal	.446	0
Egypt	.589	.018	Romania	.286	.232
El Salvador	.125	0	Russia	.607	.839
Finland	.054	0	Singapore	.054	0
France	.500	0	South Africa	.946	.089
Germany	.196	.071	Spain	.214	0
Ghana	0	.089	Sri Lanka	.054	.071
Greece	.125	.268	Sweden	.054	0
Guatemala	.464	.036	Switzerland	.089	0
Honduras	.214	.446	Taiwan	.321	0
Hungary	.429	.321	Thailand	.232	0
Iceland	0	0	Tunisia	.071	.125
India	.589	.125	Turkey	.286	.107
Indonesia	.286	.161	United Kingdom	.482	0
Ireland	.714	0	United States	.696	0
Italy	.482	0	Uruguay	.143	.143
Ivory Coast	.036	.393	Venezuela	.286	.214
Japan	.554	.054	Zambia	.161	.214
Kenya	.018	.179	Zimbabwe	.464	.286

**Table A1.** Countries and Frequency of Sanctions and External Debt Crisis

	Obs.	Frequency
Not under sanction	2,243	72.57%
Under Sanction	848	27.43%
Total	3,091	100%

 Table A2.
 Sanction Status

Table A3.	Debt	Crisis	Status
-----------	------	--------	--------

	Not in an external debt crisis	In an external debt crisis	Total
Not in a domestic or	2,612	0	2,612
external debt crisis	100%	0%	
In a domestic or	35	444	479
external debt crisis	7.3%	92.7%	
Total	2,647	444	3,091

Variable		Total	Not under sanction	t-test	Under sanction	Not in an external debt crisis	t-test	In an external debt crisis
Gdp pc	mean	1.120	.985	* * *	1.474	1.147	*	1.046
	s.d.	(.904)	(.716)		(1.204)	(.963)		(.711)
Gdp pc growth	mean	.058	.060		.053	.073	* * *	.016
	s.d.	(.138)	(.134)		(.148)	(.123)		(.166)
Inflation crisis	mean	.292	.266	* * *	.358	.233	* * *	.456
	s.d.	(.455)	(.442)		(.480)	(.423)		(.499)
Reserves	mean	24.651	24.613		24.750	29.918	* * *	9.966
	s.d.	(31.640)	(33.909)		(24.781)	(35.167)		(7.235)
Stock external debt	mean	61.740	59.551	*	67.473	44.078	* * *	110.974
	s.d.	(81.921)	(54.199)		(128.838)	(28.850)		(140.923)
Interests ext. debt	mean	1.166	.729	* * *	2.310	1.167		1.161
	s.d.	(2.259)	(1.295)		(3.508)	(2.237)		(2.324)
Military	mean	.074	.066	*	960.	.078		.066
	s.d.	(.263)	(.249)		(.295)	(.268)		(.248)
Elective	mean	.813	.806		.834	.806		.834
	s.d.	(.390)	(396)		(.373)	(.396)		(.373)
Riot	mean	.238	.210	* * *	.310	.224	*	.274
	s.d.	(.426)	(.407)		(.463)	(.417)		(.447)
Conflict Index	mean	15,021	13,447	* * *	19,145	14,888		15,392
	s.d.	(22,618)	(20,083)		(27, 824)	(23,678)		(19, 398)
N. Observations		981	710		271	722		259
<b>Notes</b> : Variables in a one-tailed t-te	are define st. **:	ed as discuss the difference	sed in Table 1 acce in means is	and Table 3 statistically	8. ***: the diffy significant a	ference in means is stat t 5% level in a one-tai	istically signed to the second state of the second se	nificant at 1% level : the difference in

**Table A4.** Summary Statistics (restricted sample)

		Dependen	t Variable:	
	Du	mmy for Exte	ernal Debt Cr	risis
	(1)	(2)	(3)	(4)
Sanction $t-2$	0.0171**	$0.0178^{**}$	$0.0153^{*}$	$0.0598^{**}$
	(0.0085)	(0.0086)	(0.0079)	(0.0288)
Yrs since debt crisis	0.0014*	$0.0022^{***}$	$0.0021^{***}$	0.0016
	(0.0007)	(0.0007)	(0.0007)	(0.0012)
Gdp pc $t-1$		-0.0025***	-0.0028***	0.0248
		(0.0008)	(0.0010)	(0.0389)
Gdp pc growth $t-1$		-0.0742	-0.0831	-0.1285
		(0.0575)	(0.0583)	(0.1277)
No inflation crisis $t-2$			-0.0193	-0.0102
			(0.0172)	(0.0324)
Elective			-0.0179	0.0311
			(0.0184)	(0.0635)
Military			0.0352	0.1077
			(0.0512)	(0.0908)
Population			-0.0003***	0.0011
			(0.0001)	(0.0007)
Reserves			· · · ·	-0.0002
				(0.0005)
Stock debt				0.0026**
				(0.0011)
Interest ext. payments $t-1$				-0.0082
~ ~				(0.0116)
Country fixed effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes
Observations	2130	2130	2130	476

Table A5. Regression of sanctions (in previous two years) on external default.

**Notes**: Variables are defined as explained in Table 1 and 3. "Sanction t - 2" is a dummy variable that takes value 1 if the country was hit by at least one sanction in the last two years. Observations are excluded if the country was under a banking crisis, a domestic debt crisis or and external debt crisis in any of the three previous years. Standard errors (in parenthesis) are clustered at the country level.

		Dependent	Variable:	
	Dur	nmy for Exte	rnal Debt Cri	isis
	(1)	(2)	(3)	
Sanction $t-3$	0.0180*	$0.0187^{**}$	$0.0167^{*}$	$0.0538^{*}$
	(0.0093)	(0.0094)	(0.0085)	(0.0296)
Yrs last debt crisis	0.0024***	$0.0034^{***}$	$0.0034^{***}$	$0.0034^{**}$
	(0.0006)	(0.0007)	(0.0006)	(0.0013)
Gdp pc $t-1$		-0.0030***	-0.0034***	-0.0053
		(0.0009)	(0.0011)	(0.0356)
Gdp pc growth $t-1$		-0.0680	-0.0758	-0.1316
		(0.0595)	(0.0607)	(0.1138)
No inflation crisis $t-2$			-0.0211	-0.0216
			(0.0193)	(0.0397)
Elective			-0.0240	0.0224
			(0.0193)	(0.0565)
Military			0.0205	0.0475
			(0.0511)	(0.0830)
Population			-0.0003***	0.0006
			(0.0001)	(0.0006)
Reserves				0.0001
				(0.0005)
Stock external debt				$0.0028^{**}$
				(0.0013)
Interest ext. payments $t-1$				-0.0032
				(0.0180)
Country fixed effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes
Observations	2007	2007	2007	445

Table A6. Regression of sanctions (in previous three years) on external default.

**Notes**: Variables are defined as explained in Table 1. "Sanction t-3" is a dummy variable that takes value 1 if the country was hit by at least one sanction in the last three years. Observations are excluded if the country was under a banking crisis, a domestic debt crisis or and external debt crisis in any of the four previous years. Standard errors (in parenthesis) are clustered at the country level.

	Dependent Variable:			
	Dummy for External Debt Crisis			
	(1)	(2)	(3)	(4)
Sanction $t-4$	0.0196**	0.0200**	$0.0173^{**}$	$0.0522^{*}$
	(0.0085)	(0.0086)	(0.0076)	(0.0271)
Yrs since debt crisis	0.0027***	$0.0038^{***}$	$0.0036^{***}$	$0.0038^{*}$
	(0.0007)	(0.0008)	(0.0007)	(0.0020)
Gdp pc $t-1$		-0.0031***	-0.0034***	-0.0124
		(0.0010)	(0.0012)	(0.0471)
Gdp pc growth $t-1$		-0.0724	-0.0829	-0.1125
		(0.0614)	(0.0623)	(0.1251)
No inflation crisis $t-2$			-0.0340*	-0.0234
			(0.0191)	(0.0412)
Elective			-0.0178	0.0211
			(0.0189)	(0.0605)
Military			0.0545	0.0503
			(0.0394)	(0.0840)
Population			-0.0003***	0.0006
			(0.0001)	(0.0006)
Reserves				0.0001
				(0.0005)
Stock ext. debt				$0.0032^{**}$
				(0.0015)
Interest ext. payments $t-1$				0.0005
				(0.0168)
Country fixed effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes
Observations	1893	1893	1893	421

Table A7. Regression of sanctions (in previous four years) on external default.

**Notes**: Variables are defined as explained in Table 1. "Sanction t - 4" is a dummy variable that takes value 1 if the country was hit by at least one sanction in the last four years. Observations are excluded if the country was under a banking crisis, a domestic debt crisis or and external debt crisis in any of the five previous years. Standard errors (in parenthesis) are clustered at the country level.

	Dep. Variable:		Dep. Variable:	
	Sanction $t + 1$ .		Sanction $t+2$ .	
	(1)	(2)	(3)	(4)
Ext. debt crisis	0.0660	0.0646	0.0563	0.0614
	(0.0500)	(0.0630)	(0.0666)	(0.1030)
Gdp pc $t-1$	0.0062	0.1080	0.0064	0.0797
	(0.0058)	(0.0948)	(0.0057)	(0.1027)
Gdp pc growth $t-1$	0.0248	$0.2770^{*}$	-0.0296	0.1507
	(0.1188)	(0.1545)	(0.1121)	(0.1429)
Yrs since debt crisis	-0.0019	-0.0014	-0.0024	0.0000
	(0.0035)	(0.0045)	(0.0035)	(0.0048)
No inflation crisis $t-2$	-0.0420	0.0083	-0.0519	-0.0174
	(0.0393)	(0.0669)	(0.0354)	(0.0631)
Elective	-0.0697	0.0776	-0.0186	$0.1684^{**}$
	(0.0687)	(0.1150)	(0.0601)	(0.0679)
Military	0.1865**	$0.2835^{**}$	$0.1416^{*}$	0.3213***
	(0.0931)	(0.1357)	(0.0779)	(0.1136)
Population	0.0006*	-0.0002	$0.0006^{*}$	0.0005
	(0.0003)	(0.0009)	(0.0003)	(0.0012)
Reserves		-0.0013*		-0.0014*
		(0.0006)		(0.0007)
Stock ext. debt		-0.0032*		-0.0016
		(0.0016)		(0.0018)
Interest ext. payments $t-1$		0.0451***		0.0362
		(0.0159)		(0.0247)
Country fixed effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes
Observations	2268	520	2268	520

Table A8. Regression of external default on sanctions in following years.

**Notes**: Variables are defined as explained in Table 1 and Table 3. "Sanction t + 1" dummy variable that takes value 1 if the country is targeted by at least one sanction in year t + 1. "Sanction t + 2" dummy variable that takes value 1 if the country is targeted by at least one sanction in year t + 2. Observations are excluded if the country was under a banking crisis, a domestic debt crisis or and external debt crisis in any of the five previous years. Standard errors (in parenthesis) are clustered at the country level.

	Dependent Variable: Riot		Dependent Variable:		
			Conflict Index		
	(1)	(2)	(3)	(4)	
Ext. debt crisis $t-3$	-0.0030	-0.0013	-324.6752	-511.6584	
	(0.0423)	(0.0738)	(319.7875)	(665.7826)	
Sanction $t-3$	-0.0240	0.0021	90.4739	98.4893	
	(0.0232)	(0.0455)	(177.4841)	(295.2822)	
Interaction2	-0.1075	$-0.1739^{**}$	-666.8819	$-1081.5405^{**}$	
	(0.0781)	(0.0853)	(466.1399)	(532.3591)	
Gdp pc $t-1$	-0.0031	-0.0346	-48.8155*	117.6046	
	(0.0046)	(0.0571)	(28.4170)	(400.2131)	
Gdp pc growth $t-1$	-0.1875**	-0.1461	-885.4994**	-1301.7373**	
	(0.0715)	(0.1413)	(338.4249)	(617.6172)	
Elective	-0.0269	-0.1064	-737.9624**	-884.0702**	
	(0.0541)	(0.0921)	(287.8510)	(411.1523)	
Military	0.0014	-0.0743	-183.9416	-477.7758	
	(0.0898)	(0.1486)	(360.5573)	(418.5811)	
n coups	-0.0097	-0.0194	-194.0943	-166.2497	
	(0.0186)	(0.0513)	(134.2303)	(263.7016)	
No inflation crisis $t-2$	-0.0290	0.0363	-285.3877*	27.4720	
	(0.0350)	(0.0434)	(148.6120)	(185.2134)	
Yrs since ext. debt crisis	-0.0011	-0.0023	-13.4669	-26.5585	
	(0.0019)	(0.0026)	(13.4716)	(26.4839)	
Yrs. since executive change	-0.0007	0.0013	-33.8540***	-39.2312**	
	(0.0022)	(0.0031)	(10.0416)	(14.8119)	
Population	-0.0006	-0.0006	-4.1739	-2.8632	
	(0.0005)	(0.0008)	(2.8774)	(5.0324)	
Reserves		-0.0008		0.9073	
		(0.0010)		(2.6075)	
Stock ext. debt		0.0008***		$2.1437^{**}$	
		(0.0002)		(0.8057)	
Interest ext. payments		0.0192		100.2130	
		(0.0174)		(113.1972)	
Country fixed effects	Yes	Yes	Yes	Yes	
Year fixed effects	Yes	Yes	Yes	Yes	
Observations	2756	978	2755	977	

## Table A9. Sanctions, External Debt Crisis and Political Unrest (greater time lag)

**Notes**: Variables are defined as described in Table 1 and Table 3. The variable "Interaction2" is a dummy for the fact that equals 1 if, in the last three years, the country was both under sanction and in an external debt crisis provided that the former status occurred before the latter. Standard errors (in parenthesis) are clustered at the country level.